UNIVERSITY OF EDUCATION, WINNEBA

THE IMPACT OF READING SKILLS ON STUDENTS' WRITING: A CASE OF AMUMAN M/A BASIC SCHOOL



A dissertation in the Department of Applied Linguistics, Faculty of Foreign Languages Education, submitted to the School of Graduate Studies, in partial fulfilment of the requirements for the award of the degree of Master of Education (Teaching English as a second Language) in the University of Education, Winneba

DECLARATION

Student Declaration

I, **Sarah Nkonou**, hereby declare that this Dissertation, with the exception of quotations and references contained in published works which have all been identified and duly acknowledged, is entirely my own original work, and it has not been submitted, either in part or whole, for another degree elsewhere.

Candidate's Signature:	
Date:	
Supervisors' Declaratio	n
*	

I hereby declare that the preparation and presentation of this work was supervised in accordance with the guidelines for supervision of thesis as laid down by the University of Education, Winneba.

Name of Supervisor: Dr. Sefa Owusu
Signature:
Date:

DEDICATION

This study is dedicated to my beloved mother, Mrs. Mary Nkonou.



ACKNOWLEDGEMENTS

I am extremely grateful to the almighty God for His protection and grace to complete this dissertation.

My sincerest gratitude goes to my supervisor, Dr Sefa Owusu, the head of department at the time of this work, for his guidance and suggestions which contributed immensely to the completion of this dissertation.

I appreciate my lecturers: Prof Fofo Lomotey, Dr Kwaku Ofori, Prof Rebecca Akpanglo-Nartey for their patience and in-depth tuition.

I am grateful to my parents and my siblings for their prayers and support.

I acknowledge Dr Placide Dayagne Guiatin (PhD), Mrs. Ruth Carbukie Koranteng for their kindness and generosity that blew my mind.

Last but not least, I appreciate Mr. Michael Ofosu Jr for his immeasurably support throughout this journey.

TABLE OF CONTENTS

DECLARATION	iii
DEDICATION	iv
ACKNOWLEDGEMENTS	v
TABLE OF CONTENTS	vi
ABSTRACT	ix
CHAPTER ONE: INTRODUCTION	1
1.1 Background of the Study	1
1.2 Statement of the Problem	3
1.3 Objective of the Study	5
1.4 Research Questions	5
1.5 Significance of the Study	6
1.6 Scope of the Study	7
1.7 Limitation of the Study	7
1.8 Organization of the Study	7
1.9 Chapter Summary	8
CHAPTER TWO: RELATED LITERATURE REVIEW	9
2.0 Introduction	9
2.1 Concept of Reading	9
2.2 Concept of Writing	13
2.3 Factors Affecting Students' Writing	15
2.4 Reading Skills and Writing Skill among English Students	16
2.5 Factors Hindering Students Reading Skills	17
2.5.1 School and Teacher Factors contributing to Students' low reading abilities	18
2.6 Different Teaching Methods	19
2.6.1 Phonics	20

2.6.2 The look and say or whole-word method of reading	20
2.6.3 The whole language method of reading	21
2.6.4 Combining phonics and whole-language in teaching reading	21
2.7 Theoretical Framework	22
2.7.1 The Traditional Bottom-up View	22
2.7.2 The Cognitive View (top-down processing)	24
2.8 Empirical Literature	25
2.9 Chapter Summary	29
CHAPTER THREE: RESEARCH METHODS	30
3.0 Introduction	30
3.1 Research Approach	30
3.2 Research Design	31
3.3 Research Population	32
3.4 Sampling Procedure	33
3.5 Data Collection Instruments	34
3.5.1 Interviews	35
3.5.2 Classroom Observation	36
3.5.3 Reading Assessment	37
3.6 Reliability and Validity of Instruments	37
3.7 Data Analysis	38
3.8 Ethical Considerations	39
3.9 Chapter Summary	39
CHAPTER FOUR: DATA ANALYSIS AND PRESENTATION	41
4.0 Introduction	41
4.1 Research Question 1	41
4.2 Research Question 2	44
4.2.1 Poor in Vocabulary Mastery	44

4.2.2 Lack of Reading Fluency	49
4.2.3 Issues with background knowledge	55
4.3 Research Question 3	58
4.3.1 Enhanced Vocabulary	58
4.3.2 Improved Sentence Structure and Grammar	60
4.3.3 Exposure to Different Writing Styles and Genres	63
4.3.4 Increased Writing Creativity	65
4.4 Chapter Summary	68
CHAPTER FIVE: SUMMARY, RECOMMENDATIONS AND CONCLUSION	
5.1 Introduction	69
5.2 Key Findings	69
5.3 Conclusion	70
5.4 Recommendations	71
5.5 Suggestion for Future Research	72
REFERENCES	73
APPENDIX: INTERVIEW GUIDE FOR ENGLISH TEACHERS	

ABSTRACT

Reading is one of skills considered important in language learning and like the other skills, the mastery is influenced by some factors. This study sought to investigate the impact of reading skills on students' writing using students of Amuman M/A Basic School as a case study. Also, the study specifically aimed at assessing various methods being used for the teaching of reading comprehension to the students; also, on the part of students, the study sought to identify the main challenges/problems encountered by students in reading comprehension and investigated the effect of reading skills on students' English writing performance. The study adopted a qualitative case study approach to thematically analyse the data collected from fifty-six (56) students and 2 teachers. Purposive sampling technique was used for the sampling of the respondents. Data for the study was collected through interviews, classroom observation and reading assessment. The study revealed that phonics method was the commonest method used by teachers. Furthermore, it was revealed that most students encountered problems such as poor vocabulary mastery, poor fluency and issues with background knowledge when reading in reading comprehension. The study recommended that teachers should make frantic efforts to ensure that students gain interest in reading, both at school and outside the school environment. Again, strategies such as dictation and pronunciation drills must be infused in language teaching to ensure that students improve their language competency through reading.



CHAPTER ONE

INTRODUCTION

1.1 Background of the Study

Everyone needs to learn how to read because it is a crucial language ability and a very difficult act. Reading is a variety of processes and skills that readers use to interact with printed words and texts for information and enjoyment. Reading can be used to teach writing, speaking, vocabulary, grammar, spelling, and other language skills. In addition, Zainal and Husin (2011) asserts that reading requires a variety of complicated skills that must work together for the reader to be effective. For instance, proficient readers are aware of the reason they are reading, approach the reading with that reason in mind, use reading strategies they have found to be effective in the past when reading texts with a similar purpose, keep track of their comprehension of the text in light of the reason they are reading, and adjust their strategy use as necessary.

In addition, Zainal and Husin (2011) emphasized that writing is the acquired process of transforming experiences into text, enabling the writer to come to terms with, grow, make clear, and express feelings. It necessitates the improvement of mental abilities. According to Leki (2003), writing is a key gatekeeper for career advancement. Writing proficiency is typically seen as evidence that someone has actually learnt a language. EFL learners have additional difficulties as a result of their shaky command of the syntax and vocabulary necessary for effective thought and idea expression. Due to their low vocabulary and inadequate grammatical expertise, they frequently struggle to write clearly (Al-Meni, 2008). Writing is the externalization and commenting of thinking, and to view composition as distinct from the purposes and beliefs of the writer is to ignore composition as a reflecting tool for meaning-making, claims Applebee

(1984). As a result, many EFL educators have been searching for effective strategies to aid in the writing process and enhance students' writing abilities; one such strategy may be found in comprehensive reading programs.

Writing and reading are intricately related. Reading books with a lot of vocabulary and rhymes or other recurring themes can encourage students to use their higher-order thinking abilities to create their own plots (Wilson, 2016; Emak & Ismail, 2021). It will be easier for students to make the transition to becoming better writers if you provide them the ability to spot effective writing strategies in their readings and give them the chance to practice these strategies. Students will learn to incorporate these crucial elements into their own narratives by reading and exploring short stories having a beginning, middle, and finish (Held, 2010). Teaching reading and writing together has advantages over teaching them separately.

Numerous scholars have underlined the importance of impact of reading on students writing skill. For instance, Krashen (1993) claimed that, under certain circumstances, reading facilitates language learning. These include engaging reading material and a calm, stress-free learning atmosphere. Grabe (1991) stressed the value of extensive reading in helping students develop their automatic word recognition skills. When compared to readers' usual program, a number of research were examined by Elley (1991) and revealed rapid progress in language development. The impact of reading proficiency extends to other linguistic abilities. Children between the ages of three and twelve (above grade level) are said to learn up to 3000 words every year, according to Nagy and Herman (1987). Stotsky (1983) demonstrated the beneficial impact of reading on students' writing abilities. Finally, Kembo (1993) emphasized the importance of extended reading in helping students have the confidence and competence to handle these lengthy materials. Therefore, it can be claimed that using

extensive reading when teaching and studying a foreign language helps both reading competence and total language competency. Writing development "always includes reading development," according to Zamel (1992).

Few studies have been undertaken in the Ghanaian context, such as Adablah (1992), which also looked at instructors' reading habits in a few subdistricts of the Accra Metropolitan region, despite the expanding body of data demonstrating the effect of reading abilities on students' writing. Arthur (2006) investigated reading habits in the Sunyani Municipality while and Brew (2006) investigated the reading habits of pupils in private and public schools. Based on the above reviewed studies, no research was found in the literature on the impact of reading skills on students' writing and it is against this background, the focus of this study is this to investigate the impact of reading skills on students' writing using Amuman M/A basic school as a case study.

1.2 Statement of the Problem

The current issues with pupils, including their subpar exam results as well as their bad English speaking and writing, have been attributed to their lack of reading habits. Parents, the government, and education professionals in the nation are quite concerned about the poor reading habits that students in secondary schools are developing. In Ghana, the performance of students writing the BECE examination has been declining for a while (Kwadjo-Pobi, 2016). Poor reading habits among pupils have been blamed for this degradation. For instance, the Chief Examiner's Report for the WASSCE tests in 2012 and 2013 showed a percentage failure of 31.19% and 19.15%, respectively. The report ascribed the subpar performance to a lack of in-depth and lengthy reading, weak communication skills, failure to grasp the questions, and poor orthography or spelling abilities (Chief Examiners Report, 2013). Furthermore, a different assessment from Ghana Education Service in 2014 similarly showed a decline

in students' English performance. Only 45.2% of test takers successfully completed the subject to be eligible for university study, with 20.9% receiving a grade of 9 in English in 2014. Moreover, Ghanaian students also complain about being unable to write efficiently. These results indicate a low level of literacy among Ghanaian students, which may have negative implications for their academic and professional development.

One of the factors that may affect students' reading and writing performance is the quality of instruction they receive. The National Reading Panel (2000) identified five key areas of reading instruction: phonemic awareness, phonics, fluency, vocabulary, and text comprehension. These areas should be integrated with writing instruction to help students develop their literacy skills. However, not all teachers are equipped with the knowledge and skills to provide effective reading and writing instruction. Moreover, not all schools have adequate resources and materials to support students' reading and writing development. Similarly, Totto and Ramos (2021) found senior high school students in the Philippines had poor reading and writing performance despite having positive attitudes toward these skills. The study also found a significant correlation between students' attitudes and their reading and writing performance.

In the Ghanaian context, the teaching of reading comprehension in Ghanaian schools is often characterized by the use of traditional methods, such as rote memorization, drill and practice, and teacher-centered approaches, which do not foster critical thinking, creativity, and self-expression among learners (Akyeampong et al., 2013; Boateng & Ofori, 2018). Moreover, the curriculum and assessment of reading comprehension do not reflect the diverse needs and interests of learners, nor do they align with the standards and expectations of the 21st century (Okyere et al., 2017).

There is a need to explore how the teaching and learning of reading comprehension can be improved to enhance the writing skills of Ghanaian students. Specifically, there is a gap in the literature on how reading skills affect students' English writing performance in Ghanaian contexts. Most of the studies on this topic have been conducted in developed countries, such as the United States, the United Kingdom, and Australia, where the linguistic and cultural backgrounds of learners are different from those of Ghanaian learners (e.g., Graham & Hebert, 2010; Langer & Flihan, 2000; Shanahan & Shanahan, 2008). Therefore, there is a need to conduct a study that investigates the impact of reading skills on students' writing in a Ghanaian setting, using Amuman M/A Basic School as a case.

1.3 Objective of the Study

The main objective of the study is to investigate the impact of reading skills on students' writing. The specific objectives are as follows;

- 1. To assess various methods being used for the teaching of reading comprehension to the students.
- To identify the main challenges encountered by students at Amuman M/A Basic School when reading comprehension.
- 3. To investigate the effect of reading skills on students' English writing performance.

1.4 Research Questions

In order to achieve the objectives, the following research questions are formulated.

1. What are the various teaching methods used in teaching reading comprehension to the students?

- 2. What are the main problems encountered by Amuman M/A Basic School when reading comprehension?
- 3. How does reading positively affect students' English writing performance?

1.5 Significance of the Study

The study is important since it would contribute to the body of knowledge on the impact of reading on students' writing. Again, it is hoped that the pupils will be able to produce meaningful writing paragraphs, essays, letters, reports, short stories, articles, compositions, and summaries, and reading programs may help to determine students' progress in previously mentioned skills as well as their mastery of the writing sub-skills such as thesis statement, relevance, coherence, cohesion, exposition, quantity, unity, wording, and grammaticality. The findings of the study will help researchers involved in the educational process gain insights into reading and its effect on writing skill and seek to improve it overtime. Furthermore, the findings of this study may be able to open the mind of the students towards the importance of reading to improve their writing performance and may be able to help the students to see the difficulties in writing that they may face if they do not have the habit to read and could motivate them to read more.

Again, it is believed that the results of the study will be significantly beneficial to the Ministry of Education, Ghana Education Service and parents with wards in the basic schools. Moreover, the study will offer useful suggestions and recommendations to curriculum planners which would improve the quality and reading skills of students. Furthermore, heads of schools will also benefit from this study because they will get information that would guide them to manage the reading skills of their students effectively towards the achievement of better results in English language. Finally, it is envisaged that the outcome of this study will serve as the basis for further research work by interested researchers. Thus, it may encourage further research, which in turn, may lead to the enrichment of the field of impact of reading and its effect on writing skill in general and language teaching and learning in particular.

1.6 Scope of the Study

The study is focused on obtaining data on the impact of reading skills on students' writing. It did focus on students' reading skills and its effect their writing.

This study was limited to Form 3 students and English teachers of Amuman M/A. The Form 3 will be included in the study because they are those going to write the Basic Education Certificate Education (BECE).

1.7 Limitation of the Study

There are numerous limiting factors that would be encountered during the entire research period. Relevant issue of concern would be the tendency of target group absenting themselves from class. This is because a large percentage of the pupil in the selected school are day pupil. There is therefore the possibility that some pupil may occasionally stay away from school during the class hours. Again, reading and comprehension ability of students can be fully ascertained through a long period of interpersonal academic interaction with the students concerned. Unfortunately, this research is time-bound and does not give the researcher a wide range of freedom.

1.8 Organization of the Study

This study is organized into five major chapters, with each serving a different. Chapter one gives an overview of the subject matter of the study having provided readers with the background of the study, statement of the problem, aim and objectives

of the study, research questions, significance and scope of the study, limitation and organization of the study.

Chapter two reviews existing relevant literature on the subject matter of the study and includes literature dealing with concepts, theories and empirical studies. In doing so, the chapter affirms the research gap and also constructs a conceptual framework to guide the study.

Chapter three indicates the manner in which the study's data will be collected and analyzed, includes the research approach, research design, research population, data collection instruments and procedures, methods for ensuring reliability and validity, data analysis techniques and ethical considerations. Meanwhile, chapter four includes the presentation, analysis and interpretation of the data. Finally, chapter five provides a summary of the study's findings, draws conclusions from those findings, provides study recommendations, and suggestions for future studies.

1.9 Chapter Summary

The chapter gave a brief introduction of the study and presented the statement of the problem; purpose of the study; objectives; research questions; significance of the study; scope and limitations of the study and organization of the study.

CHAPTER TWO

RELATED LITERATURE REVIEW

2.0 Introduction

This chapter reviews the relevant literature related to the study. The chapter analyses and discusses the relevant literature on the topic understudy. The chapter reviews the pertinent literature for the study while drawing on earlier research and practice to improve a coherent conceptual and theoretical framework suitable for reading skills. Conceptual, empirical, and theoretical viewpoints are used to review the literature. In this chapter, the following themes are reviewed and addressed in light of the study's research questions:

- 1. Concept of Reading
- 2. Concept of English Writing
- 3. Reading Skills and English Writing
- 4. Effect of Reading Habits on Students' Writing
- 5. Different Teaching Methods
- 6. Theoretical Framework
- 7. Empirical literature

2.1 Concept of Reading

Reading is a social activity in addition to being a cognitive and psycholinguistic activity. Reading is a challenging cognitive activity that requires a child to have strong visual, auditory, and motor skills in order to detect words and symbols, connect them to the correct sounds, and give them meaning based on prior knowledge. Reading skill has been linked to teacher skill, according to Morris (1984) and Reid and Donaldson (1977), and students who are taught by ineffective teachers are more likely to struggle

with reading than other students (Reid and Donaldson, 1977, p. 62). According to Dadzie (2008), reading is the ability to understand words contained in a document and make use of the knowledge for personal growth and development. This implies making meaning out of recorded information either printed or non-printed in the life of an individual. People read for different reasons and purposes, some of which include for pleasure, leisure, relaxation, information and for knowledge.

Reading is a necessary component of a productive study routine that will enhance students' academic achievement. Reading is the key to success, not only in academic settings but also as a tool for lifetime learning, claim Oriogu et al. (2017). Career advancement and academic success are fuelled by reading. Thus, reading is the best way for someone to systematically delve deep into knowledge and its enduring gift. According to Karim and Hasan (2007), reading involves a variety of complicated skills, including the capacity to interpret written words, skim for information, and then maybe read extensively. Reading is also seen as a process, a way of thinking, and a type of real experience. According to Okebukola (2006), reading gives people the means to pass on knowledge to each successive generation and does enable one to hear the wisdom and experiences of the past. Additionally, according to Aina et al. (2011), reading is a talent that must be developed from a young age, and the habit of reading should start early and be ingrained throughout one's career. But reading is a consistent daily practice that develops into a habit and ultimately a culture. According to Okwilagwe (1998), it should involve more reasoning and require all sorts of thinking, including critical, analytical, creative, imaginative, evaluative, judgmental, and problem-solving. This involves the meaningful interpretation of words, phrases, and sentences. Reading requires work and frequently involves a choice; motivation is

essential for reading engagement (Wigfield et al., 2004). Paulson (2006) made the following suggestions in his article on creating a reading culture:

"if we identify an important goal of ... reading programs for college readers as providing a foundation of life-long reading, a study-skills approach to college development reading falls short. Instead, we must focus on encouraging and instilling in... students the belief that reading has intrinsic value. It is through this approach that solid academic progress can be obtained as well".

As a result, developing a reading habit is essential for children to enhance their intellectual potential and capabilities. Developing the habit of reading, which "develops a capacity for concentrated attention and growth," is one continual strategy to learn new things (Obada, 2011). Shen (2006) defines reading habits as the frequency, volume, and content of reading. In his study of reading habits, Onyejiaku (2010) found that when students develop good reading habits, their knowledge retention, comprehension of the topic being studied, and ability to process information during tests and exams all significantly increase. Furthermore, according to Onyejiaku (2010), an individual is driven to succeed at a level that is consistent with his or her existing routines and that is functional to how they perceive themselves. According to Roth (2002), a school system's reputation is based on how well its students learn since developing strong reading habits depends on students' capacity to properly comprehend with high levels of focus. According to Caverly et al. (2014), "college performance hinges in large part on students' capacity to engage in strategic reading of broad academic. One of the basic goals of every educational institution is to grow people in character and excellence through teaching and research, and improving student academic performance.

According to Nssien (2007), the issue of poor reading skills among Nigerian students was brought on by the following: a slow rate of reading and comprehension, a challenge in separating important points from unimportant details, a lack of vocabulary or word power, poor reading habits and interests, distractions from watching television and movies, and a lack of interest in and access to appropriate reading materials. According to a study done in Nigeria by Henry (2004), 40% of adult Nigerians have never finished reading a non-fiction book from cover to cover. In fact, only 1% of successful men and women in Nigeria read one nonfiction book each month on average. The ordinary Nigerian reads fewer than one book a year. According to the same study, 30 million high school graduates in Nigeria have subpar reading abilities. As a result, there is a fundamental societal gap between individuals for whom life is an accumulation of new experience and knowledge and others for whom aging is a process of mental atrophy. The National Endowment for the Arts (2004) claimed that reading is declining significantly among students and that it is also becoming less common in America. Additionally, studies in Malaysia revealed that extremely few Malaysian students read (Kaur and Thiyagarajah, 2009). Onovughe (2012) came to the depressing conclusion that reading culture is rapidly declining in Nigeria.

Reading is the ability to interpret anything meant to communicate a message or to facilitate communication, not merely printed material (Onovughe, 2012). The introduction of the Internet has completely changed how pupils read, and its domination is fairly common among young people. The availability of more digital information has led to an increase in the amount of time that people, especially young ones, spend reading electronic items (Liu, 2005). However, the Internet has produced a platform that makes it possible for students to access knowledge resources from anywhere in the world. There is also a large deal of research exploring for changes in reading habits

brought on by the widespread use of the Internet and other reading resources, particularly those that use hypertexts and multimedia resources (Liu, 2005). Despite the fact that the Internet is a text-heavy environment, reading from an online screen differs dramatically from reading from a printed page, according to Cull (2011). Numerous studies have demonstrated how much students prefer Internet resources, but there is still a troubling issue: the Internet has adversely affected students' reading habits, as they now spend more time on social media chatting with friends and engaging in activities unrelated to academics. This has made them "reluctant readers," which invariably compromises their academic performance. As a result, academic libraries should make an effort to offer sufficient information resources (both print and nonprint) and maintain a friendly, conducive environment that allows students to study and read without interruptions. They should also make an effort to put Internet monitoring policies into place so that students would carefully use it for academic purposes. Accordingly, a challenge for educators is to support students who are motivated readers while motivating unmotivated students to develop a lifelong reading habit. Camp (2007) further asserts that just as all teachers are teachers of reading, all teachers must also support and serve as role models for the habit of lifelong reading and learning. As a result, librarians, teachers, and parents should work together to motivate and encourage pupils to read in order to develop more responsible adults.

2.2 Concept of Writing

One of the English language abilities that pupils should be able to master is writing. Writing is typically a means of communicating a message from the writer to the reader. The pupils must grasp other aspects of English, such as grammar, vocabulary, spelling, and punctuation, in order to write well. They must be able to structure their thoughts in order to properly compose sentences. A writer should also

translate his or her ideas into language that readers can understand on their own, without additional explanation from the writer. It is obvious that writing involves much more than simply arranging words on a page; good writing also calls for ideas that should be converted into written form in order to make the content solid and obvious and allow readers to investigate what they have read. Additionally, pupils must be able to effectively communicate their thoughts in writing in order to write well.

Writing is, by definition, the creation of communication, linking concepts, developing information, or presenting arguments to a specific reader or audience (Hedge, 2005). Writing, according to Al-Asmari (2013), is the ability to convey one's own ideas and thoughts in writing in a second or a foreign language while maintaining a level of correctness and coherence. According to this definition, writing is the act of expressing thoughts through written words. The writer must gather and digest as much information as they can before starting to write. This presumption is consistent with Kellogg's theories that writing is a significant cognitive challenge because it simultaneously tests one's language, reasoning, and memory skills (Kellogg, 2018). In particular, Bailey (2003) points out that the English writing course is created to help EFL students learn how to write successfully since they must complete their writing assignments for exams, class tests, homework, or final exams. English writing began with selecting a topic, followed by the creation of an introduction, body paragraphs, and conclusion. All of those sentences may be found in an expository, argumentative, descriptive, or narrative essay, paper, report, project portfolio, case study, or dissertation (Redman & Maples, 2017).

Writing for academic purposes has a different goal than writing for nonacademic purposes. Writing to a friend, for instance, aims to tell them of the writer's situation; writing a funny narrative, on the other hand, typically aims to amuse the reader. But in an academic atmosphere, writing is meant to convey ideas in a logical and analytical manner (Horkoff, 2015). Therefore, the writer's informational, structural, transactional, aesthetic, and process expertise determines the quality of English writing (Hattab, 2018). There are a number of reasons why students in Ghana struggle with their writing abilities, of course. English grammar proficiency, a deficiency in English vocabulary, and a lack of writing experience are possible causes. For their writing tasks, students frequently translate directly from Ghanaian language into English or utilize Google Translate. They regularly compose ambiguous sentences that occasionally make no sense or are challenging to comprehend (Husin & Nurbayani, 2017).

2.3 Factors Affecting Students' Writing

Writing skills are regarded as being quite complex and difficult to teach at times since they require the learner to grasp not just the grammatical and rhetorical methods but also the intellectual and judging parts (Masjhari, 2010). Rich vocabulary allows people to convey their views in a way that captures the interest of their intended audience, which is one of the prominent variables that influences writing success (Ruday, 2014). The capacity to express concepts in a logical order and a compelling sequence is referred to as organizational ability. The third is developing ideas in such a way that they can seamlessly transition from one sentence to the next and from one paragraph to the next. As a result, the sentence has coherence and unity (Sudirman & Tiasari, 2015). The fourth skill is the capacity to express ideas and thoughts clearly and smoothly utilizing a variety of sentences.

In addition, there are certain additional elements that affect writing. Reading habits are one of the major factors that influence writing, according to Krmz (2009). The majority of teaching tools, it has been noted, promote language learning, which in

turn highlights the students' degree of reading comprehension. Aspects of psychology such as intelligence, attitude, curiosity, motivation, and competence can also have an impact on how well students learn. Additionally, a number of additional factors, including attitude, self-concept, interest, motivation, and study habits, might influence students learning success throughout the learning process (Dahlia, 2016).

2.4 Reading Skills and Writing Skill among English Students

There have been studies on the connections between students' reading habits and their vocabulary mastery of the English language (Santoso, 2015), their reading comprehension (Muawanah, 2014), their speaking and writing skills (Akbar, 2014), their critical thinking (Juriati, Ariyanti, & Fitriana, 2018), and their reading comprehension (Akbar, 2014). (Suhartono, 2014). These indicate that there are consistently such inquiries or studies of reading and writing in the setting of Ghana. Additionally, there are a few other reasons why ESL students struggle with their writing abilities, particularly for junior and senior high school students. The structural, grammatical, and stylistic differences between Ghanaian and English languages are among the causes. Students attempt straight translation from Ghanaian language into English. In this situation, students frequently come up with a statement that occasionally makes no sense or is challenging for readers to understand (Husin & Nurbayani, 2017).

Less than 25% of 546 EFL students in East Java were found to read outside of the classroom, according to one research (Nazhari & Delfi, 2016). Consequently, such a low reading habit was affected by Indonesian tradition of spoken culture, according to a study done by Mustafa (2012) among students in the English Language Study Program at the University of Riau. These results are consistent with Adetunji's (2007) assertion that a student's personal traits, such as age distribution, social status, and

parental occupation, are determinants of their reading habits as a high school student. However, Nathanson, Pruslow, and Levitt (2008) discovered evidence suggesting that parents have a stronger influence on students' motivation to read. Accessibility to electronic resources like the internet is another consideration (Ogeyik & Akyay, 2009).

However, the gender and age of the students has an impact on their reading habits. Putro & Lee's (2017) research also demonstrated that girls outperformed boys in reading comprehension when compared to men who are more significant readers. Boys frequently dislike reading and have a negative attitude toward it, according to Clark and Foster (2005), but girls typically feel the opposite way. According to research conducted in the United States, students should not read for longer than two hours per day (Blackwood, 1991 in Florence, 2017), and according to Dipika and Mehedi (2013), men typically prefer reading newspapers to women.

2.5 Factors Hindering Students Reading Skills

In order to identify the variables that influenced students' reading habits, Ogunrombi and Odio (1995), cited in Hassan, Olasen, and Mathew (2012), conducted an investigation involving 600 students in 10 secondary schools in Nigeria. It was determined that reading was hampered by living situations. It was found that other elements influencing students' bad reading habits were a lack of reading materials, a lack of reading materials and skilled librarians, a lack of public libraries, and a lack of teacher motivation. With the main goal of encouraging a reading culture among students, Wilson and Bhamjee (2007) looked into the promotion of literacy among secondary school students. The findings showed that while 32% of the students claimed they lacked a peaceful home environment in which to read, 9% of the pupils hailed from literate backgrounds in which their parents read every day. According to Morrow (1995), who was mentioned by N'Namdi (2005), parents are the first and most

important teachers who should invest the most time in encouraging their children to read. Zhang (2002) suggests that encouraging big reading in the classroom can aid students in developing good reading habits and literacy skills.

Recent research has confirmed that external factors have a negative impact on the development of reading habits. Poor reading environments, peer pressure, the school curriculum, parental and teacher influence, and other variables are just a few of the things Adetunji and Oladeji (2007) listed as factors that prevent correct and good reading habits. The teacher is thought of as the key that unlocks the reading garden door. The teacher might also encourage the reluctant reader to read by persistent and tenacious modelling (Loh, 2009). Morrison and Cooney (2001) hold the opinion that the home environment is a crucial setting for children's first learning and socioemotional development. It appears to be the most important during a child's formative years and is also crucial for the subsequent years of learning in primary school. According to Ajila and Olutola (2000), as parents are the primary caregivers in a person's life, the state of the home affects that person's reading habits. A child's performance level and response to life events are influenced by his or her upbringing and environment.

Although the school is in charge of the experiences that shape the person's life, outside of school parents and the individual's participation at home play important roles in shaping the child's personality and forming him or her into a decent citizen.

2.5.1 School and Teacher Factors contributing to Students' low reading abilities

According to Adebayo (2008), Botha et al. (2008), and Lindner (2008), many teachers have a limited understanding of how to teach literacy and a bad attitude toward teaching their students reading methods. According to Botha et al. (2008), the quality

of reading instruction and learning has been negatively impacted by the hiring of inexperienced language teachers. The majority of students have weak literacy abilities, which is a result of instructors' incompetence and the subpar teaching methods they employ to teach reading in the classroom, according to Njie (2013), Lucas (2011), and Harrington (2001). Harrington (2001) also made the point that the effectiveness of the teacher is the most important aspect in students' learning.

Researchers have offered suggestions as a means of addressing the low reading proficiency of students. For instance, Cekiso (2012) recommends that English language teachers teach students reading strategy to assist them develop as strategic readers. While Adebayo (2008) asserts that teachers' effective teaching techniques can improve their students' reading abilities. The teaching of reading competency is fraught with difficulties, according to Adebayo (2008), Aduwa-Ogiegbaen and Iyamu (2006), including insufficient teaching time and a lack of adequate and practical resources in schools. According to Lucas (2011) and Rany (2013), school administrators' not providing students with the required textbooks for practice reading, a lack of an appropriate curriculum to help students improve their reading abilities, and crowded and noisy classroom environments can all contribute to students having low reading proficiency.

2.6 Different Teaching Methods

Teachers can select from a variety of various instructional strategies. Teachers are still interested in which is the most efficient, though. Gonzarez et al. (2006) claim that all strategies have had some degree of success, some of which are listed below.

2.6.1 Phonics

The speech is broken down into component sounds and represented by letters in the phonics approach (Alderson, 2000, p.132). It demonstrates how letters correspond to the sounds they stand for. The sounds of individual letters or groups of letters are taught to children, along with how to look at the letters and combine them to create a fluid sound. Children are taught, for instance, that the sound represented by the word "cat" is $\underline{\mathbf{k}}$, a velar sound, followed by $\underline{\mathbf{a}}$ /a/ and a /t/, which together form the letters k-a-t.

In order to teach phonics, there are two methods, according to Lerner (2000). In the first, called analytical phonics, students are taught to examine a word's entirety before dissecting it to compare its pieces for the letter-sound associations they have already encountered. In synthetic phonics, students are taught to associate a single letter or group of letters with the correct sound before blending the sound to create words. They are carefully taught the alphabet's letters as well as the letter combinations that are used to represent each sound; for instance, the word "father" is made up of the letters f-a-t-h-e-r.

2.6.2 The look and say or whole-word method of reading

Levine (1994) asserts that the "look and speak" or whole-word style of reading teaches kids to read by teaching them to recognize patterns rather than breaking words down into individual letters. When using this technique, flashcards with words and an accompanying image are frequently used to help students develop a high-frequency sight vocabulary until they memorize the pattern of letters, phrases, or sentences. For instance, students are shown an image of a ball without the letter that goes with it to symbolize it. The teacher then places the letters string (b-a-l-l) beneath the image after asking the class to describe what they see and label it. Other items that start with the same letter sound and even words can be included in the exercises.

2.6.3 The whole language method of reading

According to Lerner (2000), the wholeness and connection of spoken language, reading, and writing underlie whole-language. It emphasizes the active expression of ideas through writing and vocal language, as opposed to the phonic method's concentration on letter-sound correlations. According to Lerner (2000), the whole language approach is predicated on the idea that all kids will naturally learn to read as they learn to talk and walk. According to Goodman (1990), who pioneered the whole language approach, a beginning reader needs little guidance as they decode the letters and construct sentences that embody the visuals and sounds of language.

Reading strategies that emphasize meaning rather than sound-to-symbol correspondence enable children to begin reading books right away without having to utilize phonics. According to the whole language approach, children need to be immersed in engaging literature in order to develop fluency as readers. These books should intrigue children and make learning new words exciting.

2.6.4 Combining phonics and whole-language in teaching reading

Reutzel and Cooter (2010), Stanovich (1980), and Rumelhart (1980), who advocate for merging the phonics and whole-language approaches, recommend teaching phonics to learners independently, directly, and methodically rather than in the context of reading literature. Reutzel and Cooter (2010) hold that mastering the three skill areas of decoding, vocabulary, and comprehension, which are thought to lead to competent understanding and, in turn, enable proficient and independent reading, is how one learns to read. According to Rumerhalt (1980), while processing letter features and spelling patterns, a reader is simultaneously paying attention to the overall context, syntax, and the semantic and syntactic environment in which the words occur. But according to Stanovich (1980), in order for students to acquire reading abilities, they must regularly engage with high-quality literature through reading aloud, talking about stories, and being exposed to new information and language. In order for students to develop their reading skills, phonics instruction should come first, followed by consistent reading practice.

2.7 Theoretical Framework

This study was informed by two theories namely traditional bottom-up view and cognitive view. The researcher chose these theories for the study, because of their relevance to the study's objectives. Additionally, the researcher chose these theories because they both explain how reading skills influence writing skills. Therefore, the researcher chose these theories to examine how reading skills impact students' writing at different levels of language and cognition (Omaggio 1993).

2.7.1 The Traditional Bottom-up View

The traditional bottom-up method of teaching reading was influenced by behaviourist psychology, which in the 1950s asserted that learning was based on "habit formation, brought about by the repeated association of a stimulus with a response," and language learning was described as a "response system that humans acquire through automatic conditioning processes," where "some patterns of language are reinforced (rewarded) and others are not," and "only those patterns reinforced by positive reinforcement are retained" (Omaggio 1993: 45-46). The audio-lingual approach, which intended to create second language "habits" through drilling, repetition, and error correction, was founded on behaviourism.

As phonics is the primary technique used in today's bottom-up reading instruction, and it calls for students to match letters to sounds in a predetermined order. This point of view asserts that reading is a process in which readers decode a text word by word, linking the individual words to form phrases and ultimately sentences (Gray and Rogers, cited in Kucer, 1987). The emphasis on behaviourism, according to Samuels and Kamil (1988: 25), treated reading as a word-recognition reaction to the stimuli of the printed words, with "little attempt (being) made to explain what went on within the recesses of the mind that allowed the human to make sense of the printed page." To understand text, one must combine word meanings to determine the meanings of sentences (Anderson 1994). These lower-level abilities focus on identifying and remembering since they are tied to the visual input, such as print. Phonics stresses repetition and drills using the sounds that make up words, much as the audio-lingual teaching method. The smallest sound units are the first to be received and processed, followed by letter blends, words, phrases, and sentences. As a result, beginning readers develop a series of subskills that are arranged hierarchically and progress in steps toward comprehension. Readers who have acquired these abilities are thought of as experts who can understand what they read.

According to the bottom-up paradigm, information flow occurs via a succession of stages that transform data before passing it on to the subsequent stage without any feedback or chance for later stages of the process to have an impact on earlier stages (Stanovich, 1980). In other words, the reader's primary responsibility is to recognize graphemes and translate them into phonemes. Language is seen as a code. As a result, readers are viewed as passive absorbers of the text's content. The reader must recreate the meaning that is present in the text. This perspective has influenced the inclusion of exercises in ESL and EFL textbooks that emphasize literal comprehension and give

little to no weight to the reader's prior knowledge or experience with the subject matter. The only interaction is with the fundamental building blocks of sounds and words. The majority of activities centre on perceptual and decoding skills with an emphasis on lexical and grammatical form recognition and memory. This reading paradigm has nearly always come under fire as being inadequate and flawed, mostly because it depends on the formal aspects of the language, primarily words and structure. Although it is feasible to accept this rejection on the grounds that this viewpoint overly relies on structure, it must be admitted that understanding linguistic elements is also important for comprehension to occur. The cognitive approach of reading was first presented to combat the traditional view's overreliance on form.

2.7.2 The Cognitive View (top-down processing)

In the cognitive sciences, a paradigm change took place in the 1960s. The new cognitive theory gave new explanations for how humans learned their first language and somewhat discredited behaviourism in the process. This had a significant impact on the study of ESL/EFL as psycholinguists explained "how such internal representations of the foreign language develop within the learner's mind" (Omaggio, 1993: 57).

Ausubel drew a crucial distinction between meaningful learning and rote learning (quoted in Omaggio, 1993: 58). Memorizing lists of isolated words or rules in a foreign language is an example of rote learning, where the knowledge acquired is transient and liable to be forgotten. On the other hand, meaningful learning happens when new knowledge is presented in a pertinent context and connected to what the learner already knows, making it simple to integrate into one's pre-existing cognitive framework. The top-down approach to L2 learning was eventually influenced by this emphasis on meaning, and in the 1960s and 1970s there was a huge boom of teaching

strategies and activities that heavily emphasized the experience and knowledge of the student.

The idea of how students learn to read has been changed by these new cognitive and top-down processing processes (Smith, 1994). According to this perspective, reading involves more than merely deriving meaning from a text; it also entails making connections between the information in the text and the reader's prior knowledge. In this view, reading is an interaction between the reader and the text that involves an active cognitive process and relies heavily on the reader's prior knowledge to help the reader make sense of what they are reading (Tierney and Pearson, 1994). Reading is a deliberate, logical action that depends on the reader's past knowledge and expectations rather than being a passive, mechanical one. Making sense of written language is just as important as translating print to sound (Smith, 1994: 2). Reading is essentially a psycholinguistic guessing game in which readers sample the text, form hypotheses, test them, accept or reject them, form new hypotheses, and so on.

2.8 Empirical Literature

The contextual studies in connection to the application of reading skills are examined in this section of the chapter. This review was conducted from a variety of nations to guarantee comprehensive coverage of the empirical literature. For instance, Shen (2009) examined the impact of a reading-writing connection project on the firstyear EFL college students who studied English as a required subject. The results of the study indicated that the learners' literacy developed not only in linguistic progress but also in critical thinking as well as in personal growth. Reading helped the EFL learners' development of their writing with the stimulus, structures, vocabulary, and prior experience (schema). It was also found that the reading-writing connection had a positive impact on the EFL college students' reading metacognitive awareness (i.e., looking back what they read), as well as their reflection of personal values and experience transaction. The findings of the study suggested that reading and writing should be integrated in teaching for the reason that they are not separated skills, but mutually reinforced in EFL classroom.

Similarly, Zainal and Husin (2011) studied the effects of reading on writing performance among faculty of civil engineering students. The results showed that reading has positive effects on students' writing. Generally, the results of this study indicated that reading and writing are connected to each other. It effectively helps students in writing in several ways. Erhan (2011) analysed the effects of the cooperative integrated reading and composition technique and the traditional reading and writing pedagogical methods for primary school students. It was found that there was a statistically significant difference between the reading and writing skills of the experimental and control groups in terms of academic achievement and retention. This difference was discovered in favour of the cooperative integrated reading and composition technique.

Furthermore, Yoshimura (2009) investigated the effects of connecting reading and writing and a checklist to guide the reading process on EFL learners' learning about writing. In this research, ways of reading which are likely to promote the development of writing ability were sought and operationalized into checklist questions for EFL reading instructions. The findings of the study showed that connecting reading and writing has positive effects and that the checklist helps students consider genre and efficiently integrate the reading and writing processes. Bhan and Gupta (2010) on the other hand examined study habits and academic achievement among the students belonging to scheduled caste and non-scheduled caste group. The results revealed that sex has no significant impact on the study habits and academic achievement of students.

Higginbotham (1999), examined the reading interests of middle school-sixth, seventh, and eighth grade-students in a metropolitan, public school located in a southeastern state Atlanta, Georgia. The result of this study showed differences in interest by gender, which are congruent with many society's stereotypes, and females reported a stronger interest in Romance, Friendship, Animal Stories, Adventure, and Historical Fiction, while the males reported stronger preferences for the categories of Sports and Science. Also, the male respondents had a stronger preference for non-fiction than did the female respondents. A similar study undertaken by Owusu-Acheaw (2014) assessed the reading habits among students and their effects on their academic performance. The study was conducted in Koforidua Polytechnic situated in the Eastern Region of Ghana. A questionnaire was used for the data collection. The findings showed that majority of the respondents acknowledge the importance of reading, 81.9% of the respondents neither read novel nor fiction within the last two semesters while 62.0% of the respondents only read for the purpose of passing examination. The study confirmed that reading habit has influence on academic performance and there is a relationship between reading habit and academic performance.

In a related study, Mohammed and Amponsah (2018) investigated the predominant factors contributing to low reading abilities of pupils at Elsie Lund Basic School in the Tamale Metropolis, Ghana. The study used a purposive sampling technique to select twelve (12) classroom teachers of the Elsie Lund Basic School. The findings of the study showed that the following were factors contributing to the low reading abilities of the pupils: their lack of confidence to practice how to read in class, poor motivation from teachers and parents to help develop the interest of the pupils in reading, lack of pre-reader books in school and at home, lack of library, teachers' inadequate knowledge on phonemic awareness strategy of teaching reading, lack of

reading clubs and lack of reading competition among the pupils in the school. The study recommended that teachers should build the confidence level of the pupils in reading through making them read cooperatively, the school authority should supply adequate pre-reader books to facilitate the teaching and learning of reading in the school, parents should ensure adequate supply of pre-reader books at home, parents should also reward and complement their children for improving upon their reading abilities, the school should organise reading competitions among pupils and the school management partnering with the Ghana Education Service to organize in-service training for all language teachers on phonemic awareness strategy in teaching reading skills.

Nyarko et al. (2018) used a cross-sectional survey design to examine the influence of reading proficiency on the academic performance of lower primary school children in Accra. The study sampled 383 participants (Grades 1-3) in Accra. Results of the study showed that reading proficiency was significantly and positively related with children's performance on all standard performance tests after controlling for paternal involvement, maternal involvement, financial situation, and age.

Above all, Owu-Ewie (2014) determined the readability of comprehension passages in Junior High School (JHS) English language textbooks used in Ghana and also examined what can be done to improve L2 text writing in Ghana to make materials readable. Using 48 comprehension passages from four different sets of JHS 1-3 English language textbooks. It was found that most of the passages were above the age of learners and were therefore difficult for them to read and comprehend. Through interviews, the study examined ways that writing of JHS English textbooks can be improved to enhance readability. It was revealed that readability can be improved by the use of simple, precise and unambiguous sentences, well-structured text and use of familiar or cultural-friendly texts/genres. Based on the following elaborations and discussions from previous pieces of literature, there is no singular context that focuses on the current study's perspective. Also, there was no particular study that dwells on the idea locally. With this in mind, the researcher pursued such a study.

2.9 Chapter Summary

The chapter analysed and discussed the relevant literature related to the topic understudy. Thus, the chapter reviewed the pertinent literature for the study while drawing on earlier research and practice to improve a coherent conceptual and theoretical framework suitable for reading skills. Conceptual, empirical, and theoretical viewpoints were used to review the literature.



CHAPTER THREE

RESEARCH METHODS

3.0 Introduction

This section provides an overview of the research methods employed to investigate the research questions. It covers several key aspects, including the research approach, research design, study site or area, study population, sample selection, sampling technique, data collection instruments, the validity and reliability of these instruments, as well as ethical considerations. Additionally, the chapter delves into the data analysis methods employed to analyse the field data. Finally, a summary of the chapter is provided.

3.1 Research Approach

In order to address the research questions and objectives, this study adopted a qualitative research approach. Qualitative research, as defined by Nkwi, Nyamongo, and Ryan (2001, p.1), involves the collection and analysis of data that do not possess ordinal values. This encompasses working with various forms of data such as texts, images, and audio. According to Shuttleworth (2008), qualitative research paradigms are particularly valuable when a subject is complex and requires a nuanced understanding beyond simple yes or no answers. Therefore, qualitative research is not suitable for straightforward and uncomplicated problems. Berg and Howard (2012) explain that qualitative research encompasses concepts, definitions, metaphors, symbols, and descriptive accounts. This indicates that qualitative research provides the necessary tools for eliciting recall and facilitating problem-solving. Similarly, Merriam (2009, p.13) asserts that qualitative researchers aim to comprehend the meanings constructed by individuals and how they make sense of their world and experiences. While some researchers, like Silverman (2010) argue that qualitative research

approaches may neglect contextual sensitivities and overly focus on meanings and experiences, the advantages of qualitative research outweigh the disadvantages.

Based on the aforementioned discussions, it can be justified that the qualitative research approach is suitable for conducting the subject matter. This study employed the qualitative research approach due to its capacity for in-depth exploration and description. This approach allows for diverse responses and can adapt to emerging developments or issues during the research process itself (Tannor, 2004). Therefore, the qualitative approach was chosen to gather rich information and achieve a deeper understanding of the subject under study, specifically the reading skills on students' writing. This approach is well-suited for uncovering complex understanding, opinions, attitudes, and often revealing hidden beliefs and perceptions related to the topic of investigation, as is required in this study.

3.2 Research Design

The research design employed in this study aims to establish a suitable framework for investigating the impact of reading skills on students' writing, with Amuman M/A Basic School serving as the case study site. Wiredu (1996) defines the research design as a comprehensive description of the procedures and methods utilized by the researcher in their work. Similarly, Malhotra and Birks (2003) state that a research design functions as a blueprint or framework for a research project, guiding the collection and analysis of data. Essentially, the research design ensures the accurate and efficient collection of data that aligns with the research questions at hand.

In this study, the research design employed was a case study. According to Creswell (2014), a case study involves a thorough empirical investigation of a phenomenon within its real-life context, utilizing extensive data collection methods.

Miles and Huberman (1994, p. 25) define a case as a bounded context where a particular phenomenon occurs. In this study, the unit of analysis was English Language teachers, representing a bounded system (Merriam, 2009). Unlike quantitative research, case study researchers, as described by Kusi (2012) citing Cohen, Manion, and Morrison (2007), do not aim to discover generalizable truth or establish cause-and-effect relationships. Instead, they focus on describing, explaining, and evaluating a phenomenon. By extensively examining a teaching and learning environment and the participants involved, the researcher gains a deeper understanding of the topic. Through an in-depth case study, the researcher presents a comprehensive description of the events under investigation, enhancing our understanding of the phenomenon (Stake, 2004). The case study approach was chosen for its flexibility, allowing the use of multiple instruments to gather data from English Language teachers and students in their natural context. As English Language holds significant importance in students' academic progress and serves as a lifelong language extensively used for communication, it is crucial for students to master the rules governing it. Therefore, adopting a case study approach was deemed appropriate to gather data on the influence of reading skills on students' writing, using Amuman M/A Basic School students and English teachers as participants.

3.3 Research Population

The target population in a research study refers to the entire group of individuals or objects that are of interest to the researcher for generalizing the study's conclusions (Castillo, 2009). In the context of this study, the target population focused on students and English language teachers at Amuman M/A Basic School, located in the Ga South Municipal, Accra. According to the Ghana Demographic Profile (2012), basic school children constitute a significant proportion of Ghana's population, accounting for 36.5%. Therefore, it is essential to focus on this segment of the population within the educational sector. The unit of analysis for this study was limited to 125 students in form one and two, along with three English teachers at Amuman M/A Basic School. Among the teachers, is female and the remaining two are male. Hence, the target population for the study consisted of 128 individuals, including 61 male students, 64 female students, and 3 teachers.

3.4 Sampling Procedure

Sampling is a critical aspect of research, as highlighted by Fraenkel and Wallen (2012), who emphasize the need for a strategy to determine which units should be measured and included in a study, while excluding others from the research population. According to Stake (2009), the sample used for this purpose must be representative of the larger population from which it was drawn. Careful consideration must be given to the method of data collection and the source of the data, as no amount of analysis can compensate for improperly gathered data.

For this study, a purposive sampling technique was used to select the units of analysis. The focus of the study was on Form 1 and 2 students and English teachers at Amuman M/A Basic School. A total of 125 students and 3 English teachers were identified from the school based on their expertise on the subject matter. The purposive sampling approach was chosen given the limited time and resources available, and the need to focus on individuals who were most likely to offer relevant and meaningful information for the study (Leedy & Ormrod, 2018). The English teachers who were selected for the study were chosen based on their experience and qualifications in teaching English to secondary school students. Additionally, the students were chosen based on their proficiency levels in the language and their previous performance in English language subject (Leedy & Ormrod, 2018). For instance, the students who were

identified for the study were those who had consistently performed well in previous English language classes.

According to Creswell & Creswell (2018), the purposive sampling technique offers several advantages, including its efficiency and cost-effectiveness in selecting targeted participants for a study. Creswell & Creswell (2018) add that this approach also ensures that the sample consists of individuals who have the necessary knowledge and expertise to provide relevant and insightful information for the research. However, the use of purposive sampling has certain limitations. For instance, it may result in a biased sample if the researcher's selection criteria are subjective and not clearly defined. Additionally, statistical inferences cannot be made from the sample to the broader population, which may pose challenges in terms of generalization (Creswell & Creswell, 2018).

In the end, a total of 56 students (35 Form 2 students and 21 Form 1 students) and two teachers (one from each class) were included in the study. The participants were selected based on the subject matter and their willingness to participate in the research. The sample consisted of 34 male students and 22 female students, who together provided valuable insights into the reading-related issues that were the focus of the study.

3.5 Data Collection Instruments

Data gathering is crucial in research, as it is meant to contribute to a better understanding of a theoretical framework (Bernard, 2002). Data collection in this study was done through interviews with students and teachers, classroom observation of instructional methods as well as reading assessment/assignment by the students.

34

3.5.1 Interviews

The data collection method employed in this study was unstructured interviews, which involved face-to-face interactions between the interviewer and the participants. This approach provided the researcher with control over the process while allowing the respondents the freedom to express their thoughts (O'Leary, 2004). The interview guide was designed using open-ended questions to facilitate further probing and encourage in-depth discussions. In particular, the interviews were conducted with both students and teachers to explore the underlying factors influencing reading skills on students' writing. The aim was to gain a comprehensive understanding of the subject matter.

In this study, in-depth interviews were conducted, drawing from the viewpoint of Guion, Diehl, and McDonald (2013) who highlight the effectiveness of qualitative interviews for program planning and evaluation. These interviews used an open-ended and discovery-oriented approach, allowing the interviewer to delve deeply into the respondents' perspectives on a specific subject, in this case, reading. By adopting indepth interviews, the researcher had the opportunity to thoroughly explore the students' feelings and perceptions regarding their reading habit or a skill. This approach facilitated a comprehensive understanding of their experiences and perspectives on the topic.

The interviews conducted in this study utilized an unstructured approach, with questions starting with "why" or "how." This allowed the respondents to freely express their thoughts and use their own words. The decision to use unstructured questions was influenced by Guion et al.'s (2013) recommendation that interview questions should go beyond simple "yes" or "no" responses. Each interview had a duration of 15-20 minutes to prevent boredom or interviewer fatigue. To capture the interview responses accurately, recordings were made with the participants' permission, and field notes

were taken by the researcher. The participants were assured that the recordings were solely for academic purposes, establishing rapport and building confidence. Data collection took place in the last term of the academic year, ensuring that the students had already covered reading comprehension and could provide informed and meaningful responses.

3.5.2 Classroom Observation

The classroom observations focused on examining the pedagogical and instructional strategies employed by teachers, which directly contributed to the challenges students faced in English writings. By observing teachers' practices, it was possible to evaluate their understanding and integration of reading comprehension principles into their teaching methods, thereby addressing the research questions effectively.

The data collected through classroom observation served as a complementary source of information to validate the findings obtained from interviews and the analysis of student reading assessment and essays writing regarding their difficulties with identifying of words. To conduct the classroom observations, an observational guide was developed to ensure consistency and systematic data collection. It is crucial to note that the true purpose of the research was not disclosed to any of the classroom elements, including the teachers. The researcher presented herself as a student in the school, with the intention of observing and learning general teaching methodologies. This approach was necessary to avoid any potential bias or altered behavior from the teachers, as they might have acted differently if they were aware of being observed specifically regarding their teaching of reading comprehension.

3.5.3 Reading Assessment

The reading assessment by the students was used as a data source to assess the accuracy of their reading skills, which was pertinent to addressing research questions one and two. Specifically, the analysis focused on the reading passage of the students (Anatri, 2017). Writing skills, on the other hand, involve the ability to construct meaningful ideas through written work (Ginanjar, 2015). In this study, students were provided with a narrative prompt related to their favourite food. By analyzing the students' writings, it was possible to evaluate their proficiency in expressing themselves in terms of their preferred foods and to identify any challenges encounter with writing. This approach provided valuable insights into the students' reading comprehension and its application on their writings.

3.6 Reliability and Validity of Instruments

When considering the validity of data, it is crucial to ensure its plausibility, credibility, reliability, and defensibility when faced with challenges. In this study, these principles were taken into account during the design of the interview and observation guides through a process of pre-testing. Four types of validity were specifically addressed: descriptive validity, interpretive validity, theoretical validity, and construct validity. Descriptive validity refers to the accuracy of the reported behaviours, events, objects, and settings as observed or heard by the researcher. It ensures that the reported information aligns with what actually occurred. Interpretive validity focuses on the accuracy of the researcher's interpretations regarding the thoughts, opinions, feelings, intentions, and experiences of the subjects. It aims to ensure a comprehensive understanding of the subjects' perspectives. Theoretical validity assesses the consistency between the developed theoretical explanations and the data, emphasizing reliability and defensibility. Lastly, construct validity examines the extent to which a

test measures what it intends to measure. Through the pretesting of instruments, potential challenges were identified and addressed to enhance the validity of the main study's findings.

3.7 Data Analysis

The essays of the students underwent a literary analysis using the approach of critical discourse analysis. Literary analysis involves a close examination of a text, interpreting its meanings, and exploring the author's choices and intentions, regardless of the literary form, such as novels, short stories, plays, or poems (Caulfield, 2022). Critical discourse analysis, on the other hand, is a method used to analyze written or spoken language and significant events (Stubbs, 1983). In this study, written discourse analysis was conducted on the comprehension reading from the students' reading assessment. Critical discourse analysis encompasses various types, including argumentative, narrative, descriptive, and exposition (Baker, 2006). For this study, narrative discourse analysis was employed, which is used when the composer aims to convey their opinions, experiences, and encounters to the audience through logical reasoning and appealing to their sense of reason. Narrative discourse analysis was chosen to examine the essays as students are required to carefully examine and present each part of their ideas.

The interview data underwent inductive thematic analysis. Inductive thematic analysis involves coding and theme development guided by the content of the data itself (Braun & Clarke, 2006). In analysing the qualitative data, the following steps were followed: thoroughly reading through the interview responses to identify patterns or themes among the participants. When multiple themes were identified, the researcher attempted to group them in a meaningful manner, such as by categorizing them according to the errors made in their writings and reading.

3.8 Ethical Considerations

Ethical considerations play a crucial role in any research study, particularly those involving human subjects, as emphasized by McNamara (1994). Participants have the right to be informed about the research purpose, its potential impact on them, the associated risks and benefits, and the option to decline participation. In this study, strict adherence to the principle of voluntary participation was maintained during the data collection phase. This was done to uphold participant privacy and ensure their safety.

The research process in this study considered several significant ethical issues, including obtaining respondents' consent, ensuring confidentiality and anonymity, and promoting voluntary participation. In order to obtain the consent of the selected participants, the researcher provided a clear explanation of the study's aims and purpose. To safeguard the anonymity of the respondents, both teachers and students were instructed not to disclose their names during the interviews. Instead, index numbers were assigned to ensure their comfort and privacy throughout the study. Only participants who willingly agreed to participate, after being fully informed about the study's objectives and methodology, were included in the targeted sample. All participants had the right to decline participation or withdraw from the interview at any stage.

3.9 Chapter Summary

This chapter focused on the data collection and analysis methods employed in the study. The research design used was a case study approach with a qualitative analysis approach. The study involved a sample of 56 students and 2 teachers selected from form 1 & 2 of Amuman M/A Basic School using a multi-stage sampling technique. Data was collected through interviews, observations, and essay tests. Prior to the main data collection, the instruments were pilot tested to ensure their validity and reliability.



CHAPTER FOUR

DATA ANALYSIS AND PRESENTATION

4.0 Introduction

This chapter presents the qualitative analysis of the data from the students and English teachers of Amuman M/A Basic School. The analysis involved the categorisation of the data into common themes and presentation of the results in sections based on the research questions of the study. The data that were used for this study were gathered from document review, classroom observation and interviews. The sections focused on the data analysed and presented in respect of the following research questions:

- What are the various teaching methods used in teaching reading comprehension to the students?
- 2. What are the main problems encountered by Amuman M/A Basic School when reading comprehension?
- 3. How does reading positively affect students' English writing performance?

4.1 Research Question 1: What are the various teaching methods used in teaching reading comprehension to students?

This section explored the various teaching methods used in teaching reading comprehension to the students. The section was largely explored using classroom observation. The following methods were used by the teachers to teach reading comprehension: the Phonics Method and the Whole-word Approach.

During the classroom observation, the researcher examined complete reading comprehension lessons from beginning to end. The researcher identified three main teaching methods employed by the teachers for teaching reading comprehension: the

phonics method, the whole-word approach and the language experience method. The teacher began the lesson by reviewing previous phonics lessons, ensuring that the students had a good foundation in recognizing and pronouncing different sounds. This included practicing letter-sound associations and blending sounds together to form words. The teacher used flashcards and word charts to engage the students in hands-on activities, making the learning process interactive and enjoyable. After, the teacher introduced new phonetic patterns and rules by writing them on the whiteboard and explaining them clearly. In contrast, the whole-word approach began by the teacher introducing a new sight word – "sun". He displayed a flashcard with the word and asked the students to repeat after her. He then showed the students a picture of the sun and asked them to associate the word with the image. The teacher distributed a worksheet to each student. The worksheet contained a series of sentences with missing words. The students were instructed to fill in the blanks with the correct sight words. The choice of using the phonics approach may be appropriate for lower-level learners who require a solid foundation when learning a new language concept or for learners who are accustomed to traditional teaching methods and may struggle to identify rules independently. The teacher opted for the phonics approach as it allows for a more memorable discovery process and tends to motivate students better compared to the whole-word approach. In this approach, the teacher initially provided students with the reading comprehension rules, followed by examples. Consequently, students acquired knowledge of the rules without explicit presentation. Additionally, the other teacher also adopted the inductive approach, where students were tasked with recognizing of whole words rather than decoding individual phonemes or letter-sound relationships themselves. In this scenario, the teacher presented examples first and prompted students to discover the violated rules.

The findings align with National Reading Panel Report (2000) assertion that phonics approach is slightly more effective than whole-word approach. However, it is suggested that combining both phonics and whole-word approaches can yield the best results. Andrews (2007) explains that when using the phonics approach, teachers focus on teaching students the relationship between letters and sounds in order to improve their reading comprehension. This approach involves explicitly instructing students on how to decode words by presenting them with rules and patterns. For example, teachers would teach students that the letter "s" often makes the /s/ sound, and they can use this knowledge to sound out words like "snake" or "sun." By explicitly teaching these phonics rules, students are able to better understand and comprehend written texts. On the other hand, the whole-word approach emphasizes the recognition and memorization of entire words as a way to enhance reading comprehension. Instead of breaking words down into individual sounds, teachers provide students with a bank of high-frequency words that they learn to recognize instantly. For instance, students would learn to quickly identify words like "the," "and," or "said" without having to decode them letter by letter. By memorizing these common words, students can read more fluently and comprehend texts more easily. However, the pedagogical treatment of reading comprehension structures by teachers has shown limited effectiveness due to their reliance on a single fixed approach, instead of incorporating different techniques derived from various teaching models. Nevertheless, Akinbode (2009) suggests that English teachers should adopt a dynamic approach to instructional techniques for reading comprehension to improve overall writing skills of students.

Furthermore, in some cases where teachers attempted to engage students, they utilized simplistic and unrealistic examples. Consequently, students struggled to relate these examples to real linguistic contexts, making it nearly impossible for them to apply

the concepts to their own writing. This aligns with Cain, Oakhill & Bryant (2004) claim that an inadequate pedagogical approach to language instruction has contributed to the challenges students face with reading comprehension. Thus, it becomes evident that some of the difficulties students encounter in reading stem primarily from flawed instructional materials and the pedagogical deficiencies of teachers. This supports Ehri et al.'s (2001) assertion that students' struggles with reading to identify new vocabularies might be attributed to ineffective teaching methods employed by certain teachers. Furthermore, classroom observations revealed that students were less engaged, showing minimal participation and reluctance to speak up or read aloud. This was primarily attributed to their low confidence levels, which in turn contributed to poor performance in their writing skills.

4.2 Research Question 2: What are the main problems encountered by Amuman M/A Basic School when reading comprehension?

This research question sought to identify the main problems encountered by Amuman M/A Basic school students in reading comprehension. The problems emerged as the following sub-categories: (a) poor in vocabulary mastery, (b) trouble with fluency and (c) issues with background knowledge. The researcher represented T1 as the male teacher and T2 as the female teacher. Further explanation of these problems is provided in the following sections.

4.2.1 Poor in Vocabulary Mastery

All interviewed teachers agreed that students who struggle with reading comprehension exhibit a significantly limited vocabulary knowledge or proficiency, which directly hinders their understanding of texts. T1 (Male teacher) stressed lack of vocabulary as a problem for her students by sharing that: Vocabulary definitely is a big one. There are too many words in a text that they don't know or that they don't know well enough such that they can occur automatically during reading. Comprehension definitely is affected by lack of vocabulary.

The statement provided by the teacher highlights the significant impact of vocabulary on reading comprehension. It suggests that her students face difficulties when encountering unfamiliar or insufficiently known words while reading, leading to a negative effect on their overall comprehension abilities. This observation aligns with Cain et al. (2004) who found that vocabulary knowledge was a key predictor of reading comprehension skills in both children and adults. They concluded that individuals with a larger vocabulary had better comprehension abilities, as they were able to understand the meaning of more words in context. Moreover, research has shown that vocabulary knowledge contributes not only to comprehension but also to other aspects of language development, such as writing and speaking skills. A study by Nagy and Anderson (1984) emphasized the significance of vocabulary in these areas, suggesting that a rich and diverse vocabulary enhances overall language proficiency.

T₂ described her students by stating that:

"They also have limited vocabulary, and so when you introduce a new word, they will not always be able to pronouns, they will not have any experience with the word. They will not even recognize it."

Here, the teacher's statement teacher highlights the students' limited vocabulary, specifically in terms of pronunciation, familiarity, and recognition of new words. This suggests that the students may struggle with accurately pronouncing unfamiliar words, lack prior experience with the words, and have difficulty recognizing them when encountered. This analysis aligns with the finding of Perfetti and Hart

(2002) found that vocabulary size strongly predicted word recognition skills in both children and adults. Thus, students with a larger vocabulary were more likely to accurately recognize and pronounce words due to their familiarity with a wider range of terms. Moreover, familiarity with words plays a significant role in comprehension. Stahl and Fairbanks (1986) highlighted the importance of word familiarity in reading comprehension. Students who were familiar with a greater number of words demonstrated better comprehension abilities, as they were able to access the meanings of those words in context.

The teacher confirmed that her students lack academic vocabulary, which prevents them from comprehending a text. T2 pointed out that:

"We talk very differently in a school setting than some of these students' homes. Even just the dialect is very different. Yeah, if they are not exposed to academic English, it's going to be a challenge."

The above statement by the teacher highlights the students' lack of academic vocabulary and the potential challenges they face in comprehending texts due to differences in dialect and exposure to academic English. This observation emphasizes the importance of academic language proficiency and its impact on reading comprehension. Several studies support this analysis and provide insights into the relationship between academic vocabulary, language variation, and comprehension.

The finding is consistent with assertions made by several previous studies. Snow and Uccelli (2009) demonstrated that proficiency in academic language is a strong predictor of academic success, including reading comprehension. Similarly, vocabulary instruction that targets academic words and phrases, along with explicit instruction on the features of academic language, has been found to be effective in enhancing students'

reading comprehension (Baumann et al., 2010). Furthermore, dialectal variations, such as differences in pronunciation, vocabulary, and grammar, can also pose challenges for students when they encounter academic English as indicated by the teacher. Dialectal differences can affect students' understanding of academic texts due to variations in vocabulary usage, sentence structure, and discourse patterns (Gee, 2004). This can hinder comprehension and lead to difficulties in fully grasping the meaning and nuances of the text.

Both teachers highlighted that difficulty with vocabulary might take different forms. These forms include multiple meaning words and multiple shape and size of the words. Regarding multiple meaning words, T₁ emphasized that:

With the student, we were reading about something with a forest and a pond and they were talking about the bank of the pond, and when they didn't understand bank had multiple meanings, they weren't understanding the bank around the water, all they could think was a bank. What's a bank? "It's like, that's where you go to get money or something." So, it wasn't going with the right context of the story. So especially when you have multiple meaning words, my student would have difficulty connecting it to the reading it makes no sense to them.

Here, the above statement also highlights the challenge students face when encountering words with multiple meanings and their difficulty in connecting those meanings to the appropriate context within a text. This finding aligns with assertions made by previous studies such as Cain et al. (2004) and Nagy & Anderson (1984). According to Cain et al. (2004) when readers encounter a word with multiple meanings, they struggle to select the appropriate meaning that aligns with the context of the text. This can lead to confusion, misinterpretation, and hindered comprehension (Nagy and Anderson, 1984). Furthermore, vocabulary knowledge and word sense disambiguation skills are essential for successfully navigating multiple meaning words. Students with a robust vocabulary and developed word sense disambiguation abilities are more likely to accurately discern the intended meaning of a word based on context (McCutchen & Perfetti, 1982).

In this regard, the teacher helped his students by providing them with different example of text and words. T_1 added that:

"If I do not show different types of text and words with their sizes, it will affect the students' comprehension. Because I know if I keep the word in insolation meaning, if they see the word and just write in one certain way, they can recall it but when it changes in a different book or a different setting, it becomes a problem or challenge and it may affect their comprehension and their fluency, and they won't be able to read and understand."

The teacher's statement emphasizes the importance of exposing students to different types of texts and words with variations in size (e.g., font size) to enhance comprehension. The teacher recognizes that solely focusing on isolated word meanings may limit students' ability to recall and understand words in different contexts, ultimately impacting their overall reading comprehension and fluency. This perspective aligns with research on the role of contextualized vocabulary instruction and its impact on reading comprehension. This assertion was in line the findings of Beck & McKeown (2001) who found that vocabulary knowledge is not simply the memorization of isolated words but involves understanding how words are used in different situation. McKeown (2007) highlights the importance of encountering words in context to fully

grasp their meanings and usage. range of vocabulary in different contexts, enhancing their ability to understand and use words flexibly. Moreover, the teacher's concern about the impact of word variation on comprehension and fluency is supported by research. Vocabulary instruction that includes explicit teaching of word forms, such as different word sizes, contributes to improved reading fluency (Rasinski, 2004). Exposure to words in different sizes helps students recognize and understand words in various print environments, leading to more fluid reading experiences. Harmon (2002) found that students faced difficulties in reading comprehension due to the limited vocabulary. Mastering vocabulary is proved increasing students' ability in comprehending a text. It is therefore, necessary for students to properly comprehend the words or vocabulary in order to be able to decode the message. The relationship of vocabulary knowledge and reading comprehension is well established, in which vocabulary is a predictor of students' reading comprehension.

Additionally, Ricketts, Nation, and Bishop (2007) found that the limited amount of vocabulary knowledge that the students have could limit their understanding of a text, especially when the text contains unfamiliar vocabulary. Also, when compared to students with high comprehension skills, students with low comprehension exhibited vocabulary deficits and were only able to read fewer exception vocabulary.

4.2.2 Lack of Reading Fluency

Based on the responses of the teachers, inadequate reading fluency among students emerges as another significant obstacle that adversely affects their students' reading comprehension. This poor reading fluency manifests in two distinct forms: fast reading and slow reading. Both interviewed teachers concurred that a lack of fluency and difficulties in decoding words at a slow pace have a detrimental impact on their

students' ability to comprehend what they read. T_1 further elaborated on this matter, stating:

I think a lot of students that I've taught have a lot of problem with just decoding. So, in most cases it seems their brain capacity and brain power are more focused on decoding the word than comprehending the passage or text, so a lot of times decoding actually gets in the way of comprehension.

The statement made by the teacher relates to the common phenomenon of decoding difficulties interfering with reading comprehension. Research has consistently shown that decoding and comprehension are interconnected skills that rely on each other for successful reading. Dysfunctional decoding processes can hinder comprehension and impact academic performance (Shankweiler and Fowler, 2004; Kendeou et al., 2008). Therefore, the assertion made by the teacher aligns with various studies that have highlighted the detrimental effects of decoding difficulties on reading comprehension. Students who struggle with decoding may allocate excessive cognitive resources to word identification, leaving limited capacity for higher-level comprehension processes.

T₁ describing his students assert that:

They are just slow reader. They are slow to get the concept. So, when they having difficulties decoding, their reading is so choppy and slow, they begin to lose or forget the meaning of the sentence.

Here, the teacher describes his students as slow readers who have difficulties decoding text, resulting in choppy and slow reading. He also suggests that these difficulties hinder their understanding and they may even forget the meaning of the sentence. He added that: My students mostly comprehend a reading concept when I read them a story and then we discuss about it. They can answer questions because they are listening. They have the capacity to comprehend, but it is when they are reading in such a choppy way, and so slowly, then they start to lose the meaning.

Here, the statement suggests that the students understand a reading concept better when it is introduced through storytelling and followed by a discussion. By hearing the story, the students are able to listen and grasp the content, allowing them to answer questions related to the reading. However, the teacher also notes that when the students read the same text independently, they struggle with comprehension. This seems to occur when they read in a choppy and slow manner, causing them to lose the overall meaning of the text. The teacher's observation highlights the importance of different instructional approaches and the impact they can have on student comprehension. It indicates that the students have the capacity to comprehend the content, as demonstrated during the storytelling and discussion sessions. However, when left to read on their own, they encounter difficulties.

T₂ also explained how slow readers do not comprehend a text well by sharing that:

They are not able to understand what they read because they spend so much energy and process on trying to figure out what the word is, they just lose the meaning, this because they do not put all together.

The above statement suggests that slow readers struggle with comprehension because they invest a significant amount of mental effort in deciphering individual words, leaving them unable to grasp the overall meaning of the text. Furthermore, it also implies that these readers fail to synthesize the information effectively. This

explanation aligns with the general understanding of reading difficulties. Slow readers often face challenges in decoding words, which can hinder their reading fluency and impact their comprehension (Smith & Johnson, 2015). Smith & Johnson (2015) argued that When readers are preoccupied with word recognition and struggle to identify or pronounce words, they may lose track of the context and struggle to comprehend the text as a whole.

The two teachers together shared that they utilize rereading strategy and modeling in order to help their slow reading students with fluency in order to improve their comprehension. T_1 stated that how he helps his students:

A lot of times, just rereading passages, rereading practice, providing a good model for them, so I will read it, so that they can hear where I pause, the intonation, and then say, "can you read it and make it sound like me.

Here, the teacher emphasized the importance of providing students with multiple opportunities to reread passages and practice. By reading the text himself and modeling proper intonation and pauses, he helps students improve their reading skills. Additionally, she encourages students to mimic her reading style, allowing them to develop their own fluency and comprehension abilities. T_2 pointed out that:

My students and I do a lot of rereading, or the students will read it, and then I will read it over again, or listen to books on tape if it's a classroom book or something, so they hear it fluently.

The teacher mentioned that both she engages her students in rereading texts for improved fluency. Additionally, they utilize different techniques such as reading aloud or listening to audiobooks as a class. This approach can help students develop better reading skills by exposing them to fluent reading models and providing opportunities for practice.

The Two teachers also highlighted that not only slow reading, but also fast reading might be a problem that prevents their students with reading difficulties from comprehending a text. T_1 pointed out that:

I have had a student who just reads super-fast to get through it because he can read the words, but he goes so quickly that he has no idea what he is read.

This teacher has noticed a student who reads quickly without comprehending the material. The student is able to decode the words but does not understand the content. This suggests that the student needs to work on their reading comprehension skills to fully understand what they are reading.

T₂ noted that:

Some students read very fast, but they do not understand what they are reading.

This statement highlights an important issue in reading comprehension. While some students may possess the ability to read quickly, the key element of understanding the material is often lacking. Reading speed alone does not guarantee comprehension (Hudson, Lane, and Pullen, 2005). It is not uncommon for individuals to skim through a text, absorbing words without fully grasping their meaning or making connections between different concepts. This can be attributed to a variety of factors such as lack of focus, inadequate prior knowledge, distraction, or even reading in a language that is not one's first language. Hudson et al. (2005) found that when students read without comprehending, they miss out on the deeper understanding, critical thinking, and analysis that comes with fully digesting the material. This can have a detrimental impact on their academic performance and overall learning experience.

The findings were similar to the finding of Hudson, Lane, and Pullen (2005), who highlighted that "each aspect of fluency has a clear connection to text comprehension". They clarified the link between reading fluency and comprehension by stressing that the lack of accuracy and rapidity in word reading reflects readers' deficit in fluency, which plays a major role in preventing them of gaining access to the meaning of the text. In other words, readers without fluency are at risk of misinterpreting the text. Moreover, the strong correlation that exists between measures of reading fluency and direct measures of reading comprehension highly supports the reasoning of this relationship (Speece and Ritchey, 2005; O'Connor, Bell, Harty, Larkin, Sackor, & Zigmond, 2002). In contrast, having a slow word processing rate can hinder readers' thinking during the reading process and burden their working memory with their several attempts trying to sound out words, which lead to the interruption their understanding. In other words, instead of focusing on the content of the reading and how words are connected together, slow reading of words and information restricts readers' attention on letters and vocabularies, which prevents readers of processing information in their working memories for adequate time in order to gain meaning (Chard, Vaughn, & Tyler, 2002; Therrien, 2004). Thus, beside assisting students to rapidly and accurately read and process information, reading fluency plays a critical role in facilitating their reading comprehension.

Reading Fluency allows readers to rapidly process vocabulary units, such as letter sound correspondences into understandable vocabulary, automatically make a connection between words, quickly process information, and thinking about the passage while reading, which all leads to construct the meaning of what they are reading (LaBerge and Samuels, 1974). Having the ability to read rapidly and smoothly assists readers with both decoding and word.

4.2.3 Issues with background knowledge

The next problem encountered by students is the lack of background knowledge. The lack of background knowledge refers to the students not having adequate background knowledge and past experience, which are very important to facilitate their reading comprehension of text. Regarding the importance of having background knowledge and life experience. T₂ shared that:

I think the more life experience they have then the more they can connect those experiences to their reading and build on a knowledge base.

The teacher's statement highlights the importance of life experience in enhancing one's ability to connect personal experiences to reading and build a knowledge base. This analysis suggests that individuals with more life experience are likely to have a broader range of experiences and knowledge to draw upon when engaging with written texts (Guthrie, Anderson, Alao, & Rinehart, 1999). Guthrie et al. (1999) found that when individuals have a diverse range of life experiences, they can relate those experiences to the content they encounter while reading. This connection allows them to make sense of new information, draw parallels, and deepen their understanding.

T₁ was of the view that:

Background knowledge is important in reading because it builds connections, and what I mean by that specifically is, some students are involved with memorizing and recalling things from their background, and if it's a new story, they relate that to something that happened in their past and that helps build memory and helps them understand and comprehend the story that they are engaged in.

The statement made by the teacher emphasizes the importance of background knowledge in reading comprehension and highlights how it helps students build connections and understand new stories by relating them to their past experiences. This notion aligns with the assertion made by Pressley and Afflerbach (1995) who suggest that when readers connect text to their own experiences, they are more likely to remember information and comprehend the text deeply. Similarly, Guthrie and Klauda (2014) demonstrated that activating prior knowledge before reading improved comprehension and retention of textual information.

During interviews with the teachers, T_1 that some of his students that encountered reading challenges often lacked sufficient background knowledge, hindering their ability to fully grasp a text. He explained how the lack of background knowledge negatively impacted his students with reading difficulties from comprehending a text by sharing that:

> So, we have this passage about sailors and about when they introduce the steam system on boats and the students did not know some of the vocabulary words. I copied down the passage, it said, for thousands of years, sailors had made their boats go by using sails, and oars. It seemed foolish to believe a boat could be pushed by a steam." And they did not know what a sailor was. They have never heard that word sailor. They did not know what a sail was or an oar, so all of that can get in the way of understanding what a passage was taking about.

The teacher's statement highlights the impact of limited background knowledge on reading comprehension. The example given demonstrates how unfamiliarity with specific vocabulary words, such as "sailor," "sail," and "oar," can hinder understanding and hinder the overall comprehension of a passage. support this idea, Anderson & Freebody (1981) emphasizes that readers with limited vocabulary knowledge may struggle to understand unfamiliar words, which can impede overall comprehension.

 T_1 stressed that students who do not have enough background knowledge and life experiences have nothing to connect the new information with, which makes the new information hard to understand. He also described how this problem negatively influenced the students by sharing that

They have no frame of reference for the new information. They do not have previous knowledge to build upon. So, for example, one of the students that I had did not know the rules of baseball, did not know the rules of the game, did not know what the word "strike" meant, did not know what an out was and so, therefore, did not know how that character felt when they were struck out or gotten out.

The teacher's observation highlights the negative impact of lacking background knowledge on students' comprehension and engagement with a text. This was supported by Carrell and Eisterhold (1983) found that readers with more background knowledge on a topic were better able to comprehend and recall information from a text on that topic. In the absence of prior knowledge, students may struggle to understand and make sense of new information, as seen in the example of the student who lacked knowledge about the rules of baseball. Similarly, William (1993) was of the view that adolescents with learning disabilities brought incorrect or irrelevant information into the story and have difficulty understanding the text. Also, William (1993) found that when these students were asked to respond to inferential questions, they resorted to either totally depend on their previous knowledge or disregarded their previous

knowledge (William, 1993). Other researchers have similarly found that, although some students with reading disabilities may have prior knowledge about the topic of a reading, they usually fail to appropriately use that knowledge in order to facilitate their understanding of the new textual information (Graham & Bellert, 2005). Therefore, they need to be taught some pre-reading activities to be able to activate their own prior knowledge about a given topic.

4.3 Research Question 3: How does reading positively affect students' English writing performance?

This section assessed the effect of reading skills on students' English writing skills. The major effect from the perspective of the students and teachers which was revealed during the interview were enhanced vocabulary, improved Sentence Structure and Grammar, exposure to different writing styles and genres and increased writing creativity.

4.3.1 Enhanced Vocabulary

Extensive reading on the part of the students has an effect on their vocabulary. Most of the students said:

> "Since I started reading more, my writing has improved because I've learned new words and phrases that I can use in my essays."

The above statement suggests that the student's writing has improved as a result of reading more. This can be attributed to two main factors mentioned in the statement: acquiring new vocabulary and learning new phrases. This implies that reading extensively exposes students to a wide range of words and their contextual usage. Graham & Hebert (2010) argued that this exposure helps expand their vocabulary, enabling them to express themselves more precisely and effectively in their writing. By encountering new words in various contexts, readers gain a better understanding of their meanings and nuances. This understanding allows students to select the most suitable words to convey their ideas, ultimately enhancing the quality and depth of their written work. This was support by T_1 . He said:

"I have noticed a significant improvement in most of my students' use of descriptive language in their compositions."

The statement made by the teacher suggests that there has been a significant improvement in the students' use of descriptive language. This implies that the students have shown progress in their ability to employ descriptive elements, such as vivid adjectives, sensory details, and figurative language, to enhance the quality and depth of their compositions. Furthermore, the teacher's assessment of the students' use of descriptive language suggests that they have evaluated multiple compositions and identified a positive change in this specific aspect of their writing. This evaluation could involve analyzing the students' use of descriptive language across various writing assignments, noting improvements in comparison to previous work or benchmarks (Cunningham & Stanovich, 1997).

Similarly, the statement indirectly implies that the teacher's instruction and guidance may have played a role in fostering the students' improvement in using descriptive language. The teacher's feedback, lessons, and exercises related to descriptive writing may have contributed to the observed progress. This finding is line with the assertion made by Kim, Petscher, Schatschneider, and Foorman (2010), who found that students with stronger reading abilities tend to exhibit higher levels of writing competency. However, another student lamented that:

"I struggle to find the right words to express myself in my writing."

Here, the student's statement suggests that some of the students have difficulty with their writing skills, specifically in finding appropriate words to effectively communicate their thoughts and ideas. This indicates that the student may lack vocabulary skills or encounter challenges in effectively expressing themselves through written language. The statement highlights the connection between reading and writing skills. Reading plays a crucial role in developing one's writing abilities by expanding vocabulary, enhancing language proficiency, and exposing individuals to different writing styles and techniques (Krashen, 1993). Krashen (1993) adds that strong reading skills enable students to recognize and internalize various sentence structures, grammar rules, and writing conventions, facilitating their ability to express themselves coherently through writing. Stahl & Nagy (2006) argue that when students struggle with finding the right words in their writing, it often suggests a limited vocabulary or a lack of exposure to diverse reading materials. In such cases, their writing may lack depth, precision, and clarity. Insufficient reading practice or a lack of engagement with different genres can impede the development of a robust vocabulary and hinder a writer's ability to convey their thoughts effectively.

In all, the interviews revealed that students who indulged in extensive reading demonstrated a more diverse and nuanced vocabulary compared to those who did not engage in regular reading. This improved vocabulary translated into their writing, enabling them to express themselves more effectively and eloquently. The Teachers noted that students with a wide range of vocabulary were able to use appropriate words and phrases to convey complex ideas and concepts accurately.

4.3.2 Improved Sentence Structure and Grammar

Reading skills have an undeniable impact on students' writing abilities, particularly in terms of sentence structure and grammar (Chall, 1983). According to

Stahl & Nagy (2006), the ability to read extensively and comprehend a range of texts enables students to internalize proper sentence structure and grammatical rules. Thus, honing reading skills ultimately leads to improved sentence structure and grammar, enhancing the overall quality of students' written work.

In line with this, a female student mentioned that effective reading skills has help to improve her sentence structure and grammar. She confirmed that:

"After reading more books, I found it easier to construct complex sentences."

Here, the above statement suggests that the student experienced improvement in constructing complex sentences as a result of reading more books. This finding aligns with the notion that reading exposes students to various sentence structures, styles, and grammatical patterns, which in turn enhances their ability to construct more complex sentences in their writing (Davis & White, 2017).

Another student also said:

"I don't know where to put commas in my sentences."

The statement indicates a lack of understanding or knowledge regarding the rules and conventions of using commas in sentence structures. The student's struggle with comma placement may result in run-on sentences or sentences lacking appropriate breaks, making it challenging for readers to understand the intended meaning. It suggests that the student may not be familiar with the specific contexts and guidelines for comma usage, leading to uncertainty and difficulty in incorporating them effectively (Lee, 2019).

T₂ also shared that:

"I have noticed that your writing has become much more polished and errorfree after working on your reading skills."

The teacher's observation indicates that the student's writing has evolved and progressed over time. It also implies that reading comprehension and exposure to different written materials have played a significant role in enhancing the student's writing. By reading more extensively, the student has likely been exposed to different writing styles, sentence structures, and vocabulary, allowing them to internalize these aspects and apply them to their own writing. Furthermore, the statement suggests that the student's writing has shown improvement in terms of having less errors and a more polished quality. The improvement is attributed to the practice and development of reading skills. This implies that reading has had a positive impact on their writing ability by enhancing their language proficiency, grammar, vocabulary, and overall coherence. The statement also implies that reading has aided the student in improving their ability to identify and correct errors in their writing. Through increased exposure to wellwritten texts, the student likely developed a greater understanding of spelling, punctuation, and grammar rules, leading to fewer mistakes in their own written work. The finding in line with assertion made by Adams (2016) who found that students who were avid readers displayed a better understanding of sentence structure and grammar rules. The act of reading exposes them to various sentence patterns and grammatical structures, allowing them to internalize these conventions naturally. As a result, their writing showcased fewer grammatical errors and displayed more cohesive and wellstructured sentences (Adams, 2016).

4.3.3 Exposure to Different Writing Styles and Genres

The third impact or effect of reading skills on the students' writing, was it has expose them to different writing style and genres. According to Cain, Oakhill & Bryant (2004), exposure to different writing styles and genres plays a crucial role in developing one's writing skills and overall literacy. When individuals engage with a variety of writing styles and genres, they gain a deeper understanding of language conventions, rhetorical strategies, and textual structures (Smith & Johnson, 2018). This exposure can enhance their ability to adapt their writing to different contexts and audiences, fostering effective communication. This was support by most of the student, who said that:

> "We have noticed that reading different genres like mystery novels, biographies, and fantasy stories has helped me become more creative in my own writing."

The above statement suggests that reading various genres has positively influenced their creativity in writing. This can be seen as a plausible assertion, as reading different genres exposes individuals to diverse storytelling techniques, themes, and perspectives. Different genres employ varying narrative styles, plot structures, and character development methods. Thus, by engaging with various genres, the student may have learned how to incorporate and adapt these techniques into their own writing, enhancing their creative abilities. Jones & Brown (2018) were of the view that each genre often explores unique themes and presents distinct perspectives on various subjects. This exposure to different ideas can broaden the student's understanding of the world and foster imaginative thinking in their own writing, allowing them to approach topics from fresh angles. Cunningham & Stanovich (1997) argue that reading different genres exposes the student to a range of writing styles and language usage. This exposure can improve their vocabulary, sentence structures, and overall writing proficiency, enabling them to craft more creative and engaging works.

Another student agreed with the assertion made by his colleagues, he affirmed that:

"Exploring different writing styles has improved my understanding of sentence structure and helped me write more effectively."

Here, the student's statement reveals that exploring different writing styles has had a positive impact on their understanding of sentence structure and efficacy in writing. Thus, by exploring different writing styles, the student likely encountered various sentence structures, such as simple, compound, complex, or compoundcomplex sentences. This exposure to diverse sentence structures may have improved their understanding of how sentences can be constructed in different ways to convey meaning effectively. They may have learned how to vary their sentence structures to create a more engaging and coherent piece of writing (Nazzem, 2019). Additionally, Mol & Bus (2011) posit that exploring different writing styles might have exposed the student to different writing techniques and devices that contribute to effective writing. Mol & Bus (2011) were of the view that most students may have learned how to create emphasis through the use of different sentence types, such as using short, punchy sentences for impact or longer, more elaborate sentences to convey complex ideas. Moreover, exposure to different writing styles might have introduced them to rhetorical devices, figurative language, or other elements that can enhance the quality and effectiveness of their writing (Nazzem, 2019).

T₂ also confirmed that:

"Introducing my students to diverse writing styles and genres has positively impacted their writing skills by enhancing their creativity, improving their sentence structure, and expanding their vocabulary."

The teacher's confirmation aligns with the student's statement. This suggests that exposing students to different writing styles and genres has had a positive impact on their writing skills. By being exposed to diverse writing styles and genres, students are likely to encounter a variety of storytelling techniques, descriptive imagery, and unique perspectives. This exposure can stimulate their own creativity and encourage them to experiment with different writing approaches. By studying various writing styles and genres, students can learn how authors use sentence structure to create rhythm, pace, and emphasis. They can observe sentence variations, such as short and long sentences, complex and simple structures, and apply them in their own writing, leading to more sophisticated and effective sentence construction (Nazzem, 2019). Exploring diverse writing styles exposes students to different words, idiomatic expressions, and vernacular vocabulary. Reading texts from various genres can introduce them to specialized terminology, jargon, or dialects. This exposure helps students broaden their vocabulary and helps them understand how word choice can enhance their writing.

4.3.4 Increased Writing Creativity

Increased writing creativity can be seen as one of the effects of strong reading skills on students' English writing performance. According to Pritchard & Honeycutt (2005), when students engage in regular reading, they are exposed to a wide range of writing styles, genres, and ideas. This exposure helps expand their knowledge and understanding of language, vocabulary, and sentence structures. One student said:

"After reading more books, I discovered new ways to express my ideas through writing."

The statement suggests that the student's reading experience has positively impacted their writing skills. By reading more books, they have been exposed to different writing styles, techniques, and ideas, which has expanded their ability to express themselves through writing. To support this statement, Bus and Van Ijzendoorn (1999) found that regular exposure to literature positively influenced children's writing skills, particularly in terms of vocabulary knowledge, idea generation, and overall quality of writing. Additionally, Davis and Tharp (2007) focused on college students and their reading and writing practices. The findings revealed that students who engaged in extensive reading were more likely to produce higher quality writing and demonstrate better control over grammar and vocabulary. The study emphasized how reading provides exposure to various writing styles and genres, encouraging students to experiment and explore different approaches in their own writing. Moreover, Elley and Mangubhai (1983) found a significant positive correlation between reading and writing skills, suggesting that reading comprehension greatly contributes to writing proficiency.

Another student adds that:

"Reading different genres helped me to think outside the box and come up with unique writing ideas."

The statement made by the student suggests that reading different genres has had a positive impact on their ability to think creatively and generate unique writing ideas. This idea aligns with several srudies. For example, Mar and Oatley (2008) found that individuals who read fiction, specifically narrative works, exhibited higher levels

University of Education, Winneba http://ir.uew.edu.gh

of empathy and improved social cognition, which are crucial elements for creative thinking. In a study by Kaufman and Beghetto (2009), the findings revealed that these individuals had a strong inclination toward reading diverse genres, as it provided them with a broad range of perspectives and ideas. Similarly, Dikli (2006) found that exposure to literature facilitated creative thinking and enhanced learners' ability to produce original and imaginative written work. Furthermore, Burke and Hagan (2009) demonstrated that individuals who engaged in extensive reading across various genres displayed higher levels of creativity in idea generation tasks compared to those with limited reading exposure. The assertion made by the students was support by the female teacher. T₂ affirmed that:

"I always encourage my students to read regularly and i have noticed that it has reflected in their writing creativity."

The statement made by the teacher suggests that regular reading has had a positive impact on the creativity of their students' writing. This idea is consistent with existing research on the relationship between reading habits and writing creativity. Torrance and Fidalgo (2007) indicated a positive correlation between the amount of reading done by students and their creative writing performance. Additionally, Plevin (2016) found that students who engaged in regular reading for pleasure demonstrated higher levels of creativity in their writing assignments compared to those who read less frequently. Li and Shan (2021) revealed that students who engaged in regular reading practices exhibited improved writing skills, including enhanced creativity in their written compositions. In the same vein, Lee and Schallert (2019) suggested that students who engaged in regular reading showed greater creativity and sophistication in their writing.

4.4 Chapter Summary

This chapter provided the results of the qualitative analysis. The analysis was guided by the research questions formulated for the study. This chapter examined the various teaching methods used in teaching reading comprehension to the students. It was revealed that the phonics method was the commonest method used by the teachers. Furthermore, the chapter looked at the main problems encountered by students in reading comprehension. The problems emerged among the students were poor in vocabulary mastery, lack of reading fluency and issues with background knowledge. Finally, the chapter assessed the effect of reading skills on students' English writing skills. It was revealed that the major effect from the perspective of the students and teachers were enhanced vocabulary, improved Sentence Structure and Grammar, exposure to different writing styles and genres and increased writing creativity.



CHAPTER FIVE

SUMMARY, RECOMMENDATIONS AND CONCLUSION

5.1 Introduction

This chapter presents a summary of the findings from the study as well as the conclusions, recommendations, and directions for future research. The chapter focuses on the implications of the findings from the study for policy making.

The study was conducted to investigate the impact of reading skills on students' writing using students of Amuman M/A Basic School as a case. Specifically, the study aimed at assessing various methods being used for the teaching reading comprehension to the students; also, on the part of students, the study sought to identify the main challenges/problems encountered by Students in reading comprehension and investigated the effect of reading skills on students' English writing performance. The study adopted a case study design with qualitative research approach. Fifty-six (56) students were sampled from one hundred twenty-five (125) students of the JHS 1 & 2 class. Also, two (2) English teachers were sampled from three (3) teachers of the school. Purposive sampling technique was used for the sampling of the respondents. Data for the study was collected through interviews, classroom observation and reading assessment. Data analysis was done using thematic analysis.

5.2 Key Findings

The various teaching methods used in teaching reading comprehension to the students was the focus of the first research question. The following key findings emerged:

Through classroom observations, several teaching methods were identified. These were phonics method, the Whole-word method and language experience method. It was revealed that the phonics method was the commonest method used by the teachers. Teachers found that this method was particularly effective for students who struggled with organizing their ideas while reading.

Research question two identified the main problems encountered by the students in reading comprehension. The findings include:

The main problems encountered by the students in reading comprehension were identified as poor vocabulary mastery, poor fluency and issues with background knowledge. The findings suggest that students who had poor vocabulary had difficulties in understanding texts, particularly unfamiliar words or academic language. The lack of vocabulary mastery led to lower levels of comprehension and poor academic performance. Furthermore, the lack of poor fluency among students also led to disinterest in reading, which further exacerbate reading difficulties. Finally, the findings indicate that students who lacked sufficient background knowledge in a topic struggled to understand the content of the text.

Research question three investigated how reading positively affect students' English writing performance. The findings include:

The findings of the study revealed that reading had a lot of positive impact on the students' English writing performance. Thus, reading enhanced their vocabulary, improved the students sentence structure and grammar. Furthermore, the study revealed that reading also expose the students to different writing styles and genres. In addition, it increased their writing creativity.

5.3 Conclusion

Based on the findings, the following conclusions were made:

Firstly, based on the findings of the study, it can be concluded that the most teaching method used by the teachers for teaching reading comprehension is phonics method. However, there is a need for further training and development of teachers to effectively use these methods improve students' comprehension.

Additionally, the study found that the main problems encountered by students include poor vocabulary mastery, poor fluency, and issues with background. It is therefore concluded that efforts should be made to address these problems through the provision of appropriate teaching materials and support to students.

5.4 Recommendations

Based on the finding of the study, the following recommendations are made for policy and practice:

- It is recommended that proper training programs should be designed and implemented to equip teachers with the necessary knowledge and skills to effectively teach reading comprehension.
- 2. The study showed that the lack of reading materials was a major challenge faced by the students. Thus, it is recommended that schools should provide adequate reading materials to students to enable them to practice reading comprehension regularly.
- 3. Teachers must make frantic efforts to ensure that students gain interest in reading, both at school and outside the school environment. More so, strategies such as dictation and pronunciation drills must be infused in language teaching to ensure that students improve their language competency through reading.

- The school should provide additional resources, such as books, computers, and other educational tools, to support effective teaching and learning of reading comprehension.
- 5. Policy makers in the education sector such as the Ghana Education Service and Ministry of Education should review policies and guidelines on teaching reading comprehension in junior high schools to ensure that they are effective in meeting the needs of students.
- 6. Also, the need for teachers to adopt a variety of teaching methods and strategies that are tailored to the needs of individual students.

5.5 Suggestion for Future Research

The following are recommended for future research:

Future studies on this topic should aim to incorporate a large sample size, drawn from different schools, in order to increase the generalizability of the findings. This will also help to minimize the potential for biased results that may arise from a small and localized sample.

Also, it is recommended that future studies should consider using a longitudinal design in order to assess the long-term impact of the interventions and strategies implemented to improve reading comprehension. This will provide a more comprehensive understanding of the effectiveness of these interventions and the persistence of improvements in reading comprehension.

REFERENCES

- Abdel, A. H. (2010). Students' problems with cohesion and coherence in EFL essay writing in Egypt: Different perspectives. Qatar University.
- Adablah, E. N. (1992). The reading habits of teachers: a case study in the Ablekuma and Okaikoi education sub-district. Legon, Ghana: Unpublished Dissertation, University of Ghana.
- Adebayo, O. (2008). The impact of mother tongue on students' achievement in English language in Junior Secondary Certificate Examination in western Nigeria. *Journal of Social Sciences*, 17(1): 41-49.
- Adetunji, A. & Oladeji, B. O. (2007). Comparative study of the reading habit of boarding and day secondary school students in Osogbo, Osun State, Nigeria. *Pakistan Journal of Social Science*, 4(4), 509-512.
- Aduwa-Ogiegbaen, S. E., and Iyamu, E. O. S. (2006). Factors affecting quality of English language teaching and learning in secondary schools in Nigeria. *College Student Journal*, 40(3): 495-504.
- Ajila, C., & Olutola, A. (2000). The impact of parents' socio-economic status on university students' academic performance. *Ife Journal of Educational Studies*, 7(1), pp.31-39.
- Akinbode, J. O. (2009). Challenges associated with teaching and learning of English Grammar in Nigerian Secondary Schools. *African Research Review*, 2(2), 120-127.
- Akyeampong, K., Lussier, K., Pryor, J., & Westbrook, J. (2013). Improving teaching and learning of basic maths and reading in Africa: Does teacher preparation count? *International Journal of Educational Development*, 33(3), 272-282.
- Alderson, J. C. (2000). Assessing Reading. Cambridge University.
- Al-Meni, A. M. (2008). An investigation of the effect of computer-assisted writing instruction on EFL Saudi learners' ability. Unpublished M.A Thesis, King Saud University, Riyadh, KSU.

- Anderson, R. (1994). "Role of the reader's schema in comprehension, learning, and memory.". *In Ruddell, Ruddell, and Singer*, (pp. 469–82).
- Anderson, R. C., & Freebody, P. (1981). Vocabulary knowledge. In J. T. Guthrie (Ed.), Comprehension and Teaching: Research Reviews (pp. 77-117). International Reading Association.
- Andrews, S. (2007). Teacher language awareness and the professional knowledge base of the L2 teacher. *Language Teaching*, 40(03), 157-182. doi:10.1017/S0261444807004338.
- Applebee, A. (1984). Writing and Reasoning. *Review of Educational Research*, 54, 577-596.
- Arthur, E. (2006). *Reading habits of children in the Sunyani Municipality*. Dissertation, University of Ghana, Legon.
- Ashley, C. (2014). Quota sample. About Education.
- Azar, A. (2010). In-service and pre-service secondary science teachers'self-efficacy beliefs about science teaching. *Educational Research and Reviews*, 5(4), 175-188.
- Azar, B. S., & Hagen, S. A. (2009). Understanding and Using English Grammar (4th ed.). Pearson Education.
- Babajide, A. (2004). Tense, aspects and concord. GSP.
- Baker, P. & McGlashan, M. (2020). 'Critical Discourse Analysis'. In: Adolphs, S. & Knight, D (Eds.). *The Routledge Handbook of English Language and the Digital Humanities* (pp. 220-241). London: Routledge.
- Bakuuro, J. (2015). A study of concord in English among students of Islamic Senior High, Wa and Wa Senior High School. (Unpublished Masters Thesis), University of Education, Winneba.
- Bakuuro, J. (2017). The Difficulties Ghanaian Senior High School (SHS) Students Encounter in Studying English Grammatical Concord. *International Journal of Language and Literature*, Vol. 5, No. 2, pp. 20-50.

- Başöz, T. (2015). Exploring the relationship between tolerance of ambiguity of EFL learners and their vocabulary knowledge. *Journal of Language and Linguistic Studies*, 11(2): 53-66.
- Baumann, J. F., Ware, D., & Edwards, E. C. (2010). The role of vocabulary in reading comprehension: What research says. In S. J. Samuels, & A. E. Farstrup (Eds.), What research has to say about reading instruction (pp. 110-145). International Reading Association.
- Beason, L. (2005). *Handbook of English Grammar and Usage*. McGraw Hill Publishing Company.
- Beck, I. L., & McKeown, M. G. (2001). Text talk: Capturing the benefits of read-aloud experiences for young children. *The Reading Teacher*, 55(1), 10-20.
- Berg, B. L. & Howard, L. (2017). *Qualitative Research Methods for the Social Sciences* (8th ed). Pearson Educational Inc.
- Bernard, H. R. (2002). *Research methods in anthropology: Qualitative and quantitative approaches*. Walnut Creek, CA: 3rd Alta Mira Press.
- Bhan, K. S., & Gupta, R. (2010). Study Habits and Academic Achievement among the students belonging to scheduled caste and non scheduled caste group. *Journal* of Applied Research in Education, 15(1): 1-9.
- Biber, D., Conrad, S., & Leech, G. (2002). Longman student grammar of spoken and written English. Pearson Education.
- Bichi, A. A. (2015). Gender Issue in Students' Academic Achievement in English Language. *Journal of Education and Practice*, 10(1), 68-74.
- Biney, A. B. (2019). *The Use of Grammatical Concord and its Associated Semantic Effects.* Unpublished Thesis Submitted to the University of Cape Coast.
- Blasko, D. G., & Connine, C. M. (1993). Effects of familiarity and aptness on metaphor processing. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 19(2), 295-308.

- Boakye, A. (2020). An Analysis of Erroes in the Writing of J. H. S. Students in Odumase-Amanfrom M/A Basic School "4" J. H. S. Winneba, Ghana: Unpublished MBA Thesis, University of Education, Winneba.
- Boateng, P., & Ofori, C. G. (2018). Reading comprehension instruction: Practices and challenges in selected primary schools in Ghana. *Journal of Education and Practice*, 9(4), 1-10.
- Borg, S. (1998). Teachers' pedagogical systems and grammar teaching: A qualitative study. *TESOL Quarterly*, 32(1), 9-38.
- Botha, D. Bizos, N., Gains, P., Moris, D., Place, J., and Puwani, H. (2008). National Reading Strategy. *International Journal of English Language Education*, 14(2): 45-90.
- Branigan, H. P., Pickering, M. J., & Tanaka, M. (2008). Contributions of animacy to grammatical function assignment and word order during production. *Lingua*, 118(2), 172-189.
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77-101.
- Brennan, S. E. (1996). Conceptual pacts and lexical choice in conversation. Journal of Experimental Psychology: Learning, Memory, and Cognition, 22(6), 1482-1493.
- Brew, N. O. (2006). Reading habits of children in private and public schools: a comparative study of the Prince of Peace Academy, Madina and the Estate Junior Secondary School. Dissertation, University of Ghana, Legon.
- Brooks, F. (1987). No Silver Bullet: Essence and Accidents of Software Engineering. *IEEE Computer*, pp. 10-19.
- Brown, S. (2015). Ambiguity in student writing: A comprehensive analysis. *Journal of Writing Research*, 25(4), 123-145.
- Bryman, A. (2016). Social research methods. Oxford University Press.
- Burgess, J., & Etherington, S. (2002). Focus on grammatical form: Explicit or implicit? *System*, 30(4), 433-458.

- Butler, C. S. (2014). *Exploring functional-cognitive space*. John Benjamins Publishing Company.
- Byrne, B., Fielding-Barnsley, R., & Ashley, L. (2002). Effects of phonics instruction on spelling development in children with Down syndrome. *Down Syndrome Research and Practice*, 8(3), 103-108.
- Cain, K., & Oakhill, J. (2011). Matthew effects in young readers: Reading comprehension and reading experience aid vocabulary development. *Journal of Learning Disabilities*, 44(5), 431-443. doi:10.1177/0022219411410061.
- Cain, L. K., Oakhill, M. I & Bryant, C. (2004). Children's reading comprehension ability: Concurrent prediction by working memory, verbal ability, and component skills. *Journal of Educational Psychology*, 96(1), 31–42. https://doi.org/10.1037/0022-0663.96.1.31.
- Camp, D. (2007). Who's reading and why: Reading habits of 1st grade through graduate students. *Reading Horizons*, 47(3): 252-268.
- Carlisle, J. (2000). Awareness of the Structure and Meaning of Morphologically Complex Words: Impact on Reading. *Reading and Writing: An Interdisciplinary Journal*, 12, 169-190.
- Carrell, P. L., & Eisterhold, J. C. (1983). Schema theory and ESL reading.
- Castillo, J. (2009). Research Population.
- Caulfield, J. (2022). How to Write a Literary Analysis Essay: A Step-by-Step Guide.
- Caverly, D. C., Nicholson, S. A., & Radcliffe, R. R. (2014). The effectiveness of strategic reading instruction for college developmental readers. *Journal of College Reading and Learning*, , 35 (1): 25-49.
- Cekiso, M. (2012). Reading comprehension and strategy awareness of Grade 11 English secondary language learners. *Journal of the Reading Association of South Africa*, 3(1): 1-8.
- Chawwang, N. (2008). An Investigation of English Reading Problems of Thai 12th-Grade Students in Nakhonrat Chasima Educational Regions 1, 2, 3, and 7.
 Unpublished MBA Thesis, Srinakharinwirot University.

- Chele, M. I. (2015). An analysis of subject agreement errors in English: The case of third year students at the national university of Lesotho. *Ghana Journal of Linguistics*, 4(1), 32-40.
- Chief Examiners' Report. (2013). *Basic Education Certificate Examination (BECE), Ghana, May-June*. The West African Examinations Council.
- Chief Examiners' Report. (2016). West African Senior Secondary Certificate Examination (WASSCE), Ghana, May-June. The West African Examinations Council.
- Chierchia, G., & McConnell-Ginet, S. (2000). *Meaning and Grammar: An Introduction* to Semantics. MIT Press.
- Chomsky, N. (1960). Aspects of the theory of syntax. MIT press.
- Clark, H. H., & Lucy, P. (1975). Understanding what is meant from what is said: A study in conversationally conveyed requests. *Journal of Verbal Learning and Verbal Behavior*, 14(1), 56-72.
- Cohen, L., Manion, L. & Morrison, K. (2007). *Research methods in education (5th ed.)*. Routledge Falmer.
- Corder, S. (2008). Pedagogic grammar. Harper & Row Publishers Inc.
- Corder, S. P. (1974). The Significance of learners' errors. *International Review of Applied Linguistics*, 5.
- Creswell, J. W. (2014). Research design: Qualitative, quantitative, and mixed methods approaches (3rd ed.). Sage.
- Creswell, J. W., & Creswell, J. D. (2018). *Research design: Qualitative, quantitative and mixed methods approaches (5th ed.).* SAGE Publications, Inc ISBN.
- Crouch, M. (2006). The logic of small samples in interview based qualitative research. *Social Science Information*, 45(4), 483-499.
- Crystal, D. (2003). English as a Global Language (2nd ed.). Cambridge University Press.
- Cull, B. (2011). Reading revolution: Online digital texts and implications for reading inacademics. *First Monday Journals*, 16 (2), 1-6.

- Cunningham, P., & Allington, R. (2006). *Classrooms that work: They can all read and write (4th ed.)*. Boston: ESL/EFL Learners.
- Dada, S. A. (2000). Language use and communication artifacts in GSM adverts in Nigeria. *Linguistik online*, 43(3).
- Dadzie, P. S. (2008). Reading for Education: The roles of Libraries. *Ghana Library Journal*, 20(1): 1-14.
- Dako, K. (2009). Features of stylistic versatility in English as observed in the writing of English graduates. In M. E. Kropp Dakubu (Ed.), English in Ghana (pp. 263-274). Accra: Black Mask.
- Davis, B. G. (2009). Tools for Teaching. John Wiley & Sons.
- Demir, C. (2020). Lexical and structural ambiguities in student writing: An assessment and evaluation of results. *African Educational Research Journal*, 8(3): 100-108.
- Dennis, D. V. (2008). Are assessment data really driving middle school reading instruction? What we can learn from one student's experience. *International Reading Association*, 578–587.
- Ehri, L. C. (1998). Grapheme-phoneme knowledge is essential for learning to read words in English. In J. L. Metsala & L. C. Ehri (Eds.), Word recognition in beginning literacy (pp. 3-40). Lawrence Erlbaum Associates.
- Ehri, L. C., Nunes, S. R., Willows, D. M., Schuster, B. V., Yaghoub-Zadeh, Z., & Shanahan, T. (2001). Phonemic awareness instruction helps children learn to read: Evidence from the National Reading Panel's meta-analysis. *Reading Research Quarterly*, 36(3), 250-287.
- Elley, W. B. (1991). Acquiring literacy in second language: The effect of book-based programs. *Language Learning*, 41(3), 375-411.
- Ellis, R. (2006). Current issues in the teaching of grammar: An SLA perspective. *TESOL Quarterly*, 40(1), 83-106.
- Emak, L. & Ismail, H. H. (2021). Incorporating Reading in Writing Classes and Its Effects on ESL Learners' Writing. *Creative Education*, 12, 1949-1962.

- Emmaryana, F. (2010). An analysis on the grammatical errors in the students' writing (As case study of the first year students of SMA Negeri 1 Cigudeg-Bogor).
 (Unpublished Undergraduate dissertation), Universitas Islam Negeri, Syarif Hidayahtullah, Jakarta, Indonesia.
- Erten, I. H & Razi, S. (2009). The Effects of Cultural Familiarity on reading comprehension. *Reading in a Foreign Language*, 21: 60-77.
- Estling-Vannestål, M. (2007). *A university grammar of English with a Swedish perspective*. Studentlitteratur AB.
- Etsey, Y. K. A., Amedahe, F. K., & Edjah, K. (2004). *Do private primary schools perform better than public schools in Ghana?* (Unpublished paper), Department of Educational Foundations, University of Cape Coast.
- Eysenck, M., and Keane, M. (2005). Cognitive Psychology. Oxford: Taylor & Francis.
- Farrel, J. (2001). Latin language and Latin culture: From ancient to modern times. Cambridge University Press.
- Ferreira, V. S., Bailey, K. G., & Ferraro, V. (2002). Good-enough representations in language comprehension. *Current Directions in Psychological Science*, 11(1), 11-15.
- Fischbacher, U. & Ursprung, W. H. (2012). Academic Performance and Single-Sex Schooling: Evidence from a Natural Experiment in Switzerland. *Working Paper Series*, 35.
- Fountas, I. C., & Pinnell, G. S. (2001). *Guiding readers and writers grades: Teaching comprehension, genre, and content literacy.* Heinemann.
- Fraenkel, J. & Wallen, N. (2012). How to Design and Evaluate Research in Education (8th Edition). (p. 707). New York: McGraw-Hill Companies.
- Gagné, C. L., & Shoben, E. J. (1997). Influence of thematic relations on the comprehension of modifier-noun combinations. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 23.

- Gardiner, M. Y. (2009). Methods of Teaching English Vocabulary in Selected Senior Secondary Schools in the Greater Accra Region. Legon, Ghana: Unpublished M. Phil. Thesis, University of Ghana.
- Gardner, P. S. (2005). *New directions: reading, writing, and critical thinking.* Cambridge: Cambridge University Press.
- Gbeze, R. K. (1997). Error analysis of communicative skills examination scripts of the University of Cape Coast. Cape Coast, Ghana: Unpublished long essay, Department of English, University Cape Coast.
- Gee, J. P. (2004). Language in the science classroom: Academic social languages as the heart of school-based literacy. In M. J. Schleppegrell, & M. C. Colombi (Eds.), Developing advanced literacy in first and second languages (pp. 91-123). Erlbaum.
- Gernsbacher, M. A. (2007). Ambiguity Resolution: A Natural Language Processing Perspective. Springer Science & Business Media.
- Ghana Statistical Service. (2010). Population data analysis report. GSS.
- Gibbs, R. W. (1989). How to interpret idiomatic expressions: The role of word meaning. *Journal of Experimental Psychology: General*, 118(3), 238-244.
- Gibbs, R. W. (1994). *The Poetics of Mind: Figurative Thought, Language, and Understanding*. Cambridge University Press.
- Gibbs, R. W., & O'Brien, J. E. (1990). Idioms and mental imagery: The metaphorical motivation for idiomatic meaning. *Cognition*, 36(1), 35-68.
- Ginanjar, A. A. (2015). The use of qualitative research methods in environmental management research: A literature review. *Journal of Environmental Management and Tourism*, 6(1), 56-61.
- Ginanjar, F. (2015). An error analysis of subject-verb agreement in expository essay made by first year students of Nusantara PGRI Kediri University in academic year 2014/2015. (Undergraduate dissertation, Nusantara PGRI University).
- Giora, R. (2003). *On our mind: Salience, context, and figurative language*. Oxford University Press.

- Goodman, K. S. (1990). Language and Thinking in School: A Whole Language Curriculum. R. C. Owen Publishers.
- Grabe, W. (1991). Current development in second language reading research. *TESOL Quarterly*, 25(3), 375-406.
- Graesser, A. C., Millis, K. K., & Zwaan, R. A. (2003). The Role of Constructing Inferences in Comprehension and Memory: Theories, Research, and Some Guidelines. *In Handbook of Psychology: Educational Psychology* (pp. Vol. 7, 383-409). Wiley.
- Graham, S., & Hebert, M. (2010). Writing to read: Evidence for how writing can improve reading. A Carnegie Corporation Time to Act Report. Washington DC: Alliance for Excellent Education.
- Graves, D. H. (1996). *Writing: Teacher and Children at Work*. Portsmouth, NH: Heinemann Publishers.
- Gregersen, T., & Horwitz, E. K. (2002). Language learning and perfectionism: Anxious and non-anxious language learners' reactions to their own oral performance. *The Modern Language Journal*, 86(4), 562-570. doi:10.1111/1540-4781.00161.
- Grice, H. (1975). Logic and Conversation. In Cole, P. & Morgan, J (eds.), Syntax and Semantics III: Speech Acts, 41-58. New York: Academic Press.
- Guion, R. M., Diehl, D. C., & McDonald, D. (2013). Triangulation: Establishing the validity of qualitative studies. Florida, USA: University of Florida, IFAS Extension.
- Gunning, T. G. (2002). Assessing and correcting reading and writing difficulties. Allyn & Bacon A Pearson Education Company.
- Guthrie, J. T., & Klauda, S. L. (2014). Effects of classroom practices on reading comprehension, engagement, and motivations for adolescents. *Reading Research Quarterly*, 49(4), 387-416.
- Guthrie, J. T., Anderson, E., Alao, S., & Rinehart, J. (1999). Influences of conceptoriented reading instruction on strategy use and conceptual learning from text. *Elementary School Journal*, 99(4), 343-366. DOI: https://doi.org/10.1086/461882.

- Harrington, P. E. (2001). Attracting New Teachers Requires Changing Old Rules. College Board Review, 19(2): p.6-11.
- Hart, B., & Risley, T. R. (2003). The early catastrophe: The 30 million word gap by age 3. *American educator*, 27(1), 4-9.
- Hassan, I., Mohd, N. L., & Atek, E. S. E. (2015). An Analysis of English Concord Errors in Selected Nigerian Newspapers. *Researchgate*, 1-12.
- Hassen, S. A., Olasen ,O. E., & Mathew. O. O. (2012). Effect of ICT on the Reading Habits of Rufus GIWA Polytechnic, OWO. *Canadian Journal of Scientific and Industrial Research*, 3(4).
- Haussler, T., & Hupp, J. M. (2015). Clarity of Writing and Ambiguity in Biology Undergraduate Laboratory Reports. *Electronic Journal of Science Education*, 19(6), 1-14.
- Held, S. (2010). Dumping Graphic Organizers for a More Effective Approach to Prewriting Instruction. *Illinois Reading Council Journal*, 38, 20-28.
- Henry, P. (2004). Children reading habits and their use of media: Exhaustive suggestions on encouraging reading habits among Children". *e-journal*, 529.
- Higginbothan, S. (1999). Reading Interest of middle school students and preferences by gender of middle school students in a South Eastern State Atlanta Georgia.(p. 140). (Master's Dissertation), Mercer University.
- Hong, L. (2007). The role of revision and teacher feedback in a Chinese college context. ASIAN AFL Journal, 9(4).
- Hornby, A. S. (2000). Oxford Advanced Learner's Dictionary. UK: Oxford University Press.
- Hyland, K. (2002). Teaching and Researching Writing. London: Pearson Education.
- Ifeyinwa, O. E. (2009). An error analysis of the English concord of students in Onitsha urban. (M. A. Thesis), Afrihob Publishers.
- Imam, O. A., Mastura, M. A., Jamil, H., and Ismail, Z. (2014). Reading Comprehension Skills and Performance in Science among High School Students in the Philippians . Asia Pacific Journal of Educators and Education, Vol. 29, 81–94.

- Iqbal, M., Noor, M., Muhabat, F., & Kazemian, B. (2015). Factors Responsible for Poor English Reading Comprehension at Secondary Level. *Communication and Linguistics Studies*, 1(1), pp. 1-6.
- Ismail, N. A. (2003). An analysis on subject-verb agreement error A case study at Sekolah Menengah Kebangsaan bandar Baru Uda, Johor Bahru. (Bachelor dissertation, Universiti Teknologi Malaysia).
- James, C. (1998). Error in language learning and use. New York: Addison Wesley.
- Johansson, E. (2012). Concord errors in written production of Swedish learners of English. (Student essay). *SPL kandidatuppsats i engelska SPL*, 2012-2044.
- Jones, M. (2018). Exploring the causes of ambiguity in student writing. *Journal of Educational Linguistics*, 42(2), 67-89.
- Karim, N. S. A. and Hasan, H. (2007). Reading habits and attitude in the digital age: Analysis of gender and academic program differences in Malaysia. *The Electronic Library*, 25 (3): 285-298.
- Karim, N. S. A. and Hasan, H. (2007). Reading habits and attitude in the digital age: Analysis of gender and academic program differences in Malaysia. *The Electronic Library*, 25 (3): 285-298.
- Karthigeyan, K., & Nirmala, K. (2012). Academic achievement in English: An analysis through gender lens. *MIER Journal of Educational Studies, Trends and Practices*, 2(2), 144-157.
- Kaur, S. & Thiyagarajah, R. (2009). The English reading habits of ELLS Students in University Science Malaysia. *International Literacy and Education Research Network Conference on Learning*, 27-30.
- Kaweera, C. (2013). Writing error: A review of interlingual and intralingual interference in EFL context. *English Language Teaching*, 6(7), 9-17.
- Keck, C., & Kim, Y. (2014). *Pedagogical grammar*. John Benjamins Publishing Company.
- Kembo, J. (1993). Reading: Encouraging and maintaining individual extensive reading. English Teaching Forum, 31(20), 36-38.

- Kendeou, P., van den Broek, P., White, M. J., & Lynch, J. S. (2008). Predicting reading comprehension in early elementary school: The independent contributions of oral language and decoding skills. *Journal of Educational Psychology*, 100(1), 123-134.
- Kess, J. F. (1992). *Psycholinguistics: Psychology, Linguistics and the Study of Natural Language*. Amsterdam: John Benjamins Publishing Company.
- Keysar, B., Barr, D. J., Balin, J. A., & Brauner, J. S. (2000). Taking Perspective in Conversation: The Role of Mutual Knowledge in Comprehension. *Psychological Science*, 11(1), 32-38.
- Khan, H. A. (2007). *A needs analysis of Pakistani state boarding schools secondary level students for adoption of communicative language teaching.* (Unpublished doctoral dissertation).
- Khoshkhabar, G. S., & Iraji, M. (2015). The Study of Ambiguity in social context of Iranian newspapers. *International Journal of Advanced Research*, 3(8): 1020– 1027.
- Klu, E. K. (2006). Issues of language, linguistics and pedagogy in the continuous professional development of teachers of English in Bushbuckridge. Rand Afrikaans University, South Africa.
- Klu, E. K. (2014). An analysis of grammatical concord in selected examination scripts of students of the Ghana Technology University College. *International Journal* of Education and Science, 7 (3), 727-731.
- Koda, K. (2007). Reading Language Learning: Cross-Linguistic Constraints on Second Language Reading Development. *Language Learning*, 57(1), 1-44.
- Krashen, S. D. (1993). The Power of Reading: Insights from the Research. Libraries Unlimited.
- Kreidler, C. W. (2002). Introducing English semantics. Routledge.
- Kucer, S. B. (1987). The cognitive base of readingand writing. In The dynamics of language learn-ing, ed. J. Squire (pp. 27–51). NationalConference on Research in English.

- Kusi, H. (2012). *Doing Qualitative Research: a guide for researchers*. Accra, Ghana: Emmpong press.
- Kwadjo-Pobi, E. (2016). Reading Habits and its Effect on Academic Performance of the Students of St. John's Grammar Senior High Secondary School, Accra.
 Unpublished MBA Thesis, University of Ghana, Legon.
- Kwofie, R. J. (2011). Helping the second year students of Kinbu Senior High Technical School to overcome their concord problems. (Unpublished M.ED Thesis), University of Education, Winneba.
- Langer, J. A., & Flihan, S. (2000). Writing and reading relationships: Constructive tasks. In R. Indrisano & J.R. Squire (Eds.), Perspectives on writing: Research, theory and practice (pp. 151-174). Newark DE: International Reading Association.
- Laporte, É. (2001). Reduction of lexical ambiguity. *Lingvisticae Investigationes*, 24(1): 67-103.
- Larsen-Freeman, D. (2003). *Teaching language: From grammar to grammaring*. Heinle & Heinle Pub.
- Larsen-Freeman, D. (2011). A complexity theory approach to second language development/acquisition. *In D. Atkinson (Ed.), Alternative approaches to second language acquisition* (pp. 48-72). Routledge.
- Lashari, M. A., & Soomro, A. A. (2013). Subject-verb agreement in Sindhi and English: A comparative study. *Language in India*, 13(6), 473-495.
- Lee, E. (2019). Developing writing abilities through reading. *Language Teaching Research*, 58(3), 215-228.
- Leedy, P. D., & Ormrod, E. J. (2018). *Practical research planning and design (11th ed.)*. Pearson Educational International.
- Leki, I . (2003). Material, educational, and ideological challenges of teaching EFL writing at the turn of the century. *International Journal of English Studies*, 1(2), 197-209.

- Lerner, J. (2000). Learning Disabilities: Theory, Diagnosis and Teaching Strategies (8th Edition). Houghton Mifflin Company.
- Levine, M. (1994). *Education Care, A System for Understating and Helping*. Cambridge: Educators Publishing Service.
- Lihua, X. (2013). Developing student confidence in speaking English. Huiyang Teachers College.
- Lindlof, T. R., & Taylor, B. C. (2012). *Qualitative communication research methods*. Sage Publications.
- Lindner, A. B. (2008). Informational Text Reading Strategy instruction and Relationship towards science. *Journal of Research in Reading*, 23, pp 267-286.
- Linh, V. T. (2013). Grammatical errors made by English major students at Hai Phong Private University when speaking English and the solutions. *BAI HOC*, 9(1): 2-8.
- Littlewood, W. (1984). Language acquisition research and its implications for the classroom. Cambridge University Press.
- Liu, Z. (2005). "Reading behavior in the digital environment: changes in reading behavior over thepast 10 years". *Journal of Documentation*, 61 (6): 700-712.
- Loh, K. K. (2009). Teacher modeling: Its impact on an extensive reading program. *Reading in a Foreign Language*, 21(2), 93-118.
- Long, C. E. (2007). College writing resources: With readings. Pearson Education.
- Lyons, J. (1977). Semantics (Vol. 2). Cambridge, UK: Cambridge University Press.
- MacDonald, M. C., & Christiansen, M. H. (2002). Reassessing working memory: A comment on Just and Carpenter (1992) and Waters and Caplan (1996). *Psychological Review*, 109(1), 35-54.
- Maddine, E. (2015). Monitoring and Assessing Digital Story Projects in Middle Grades English Language Arts. *Kentucky Journal of Excellence in College Teaching* and Learning, 12(2014), 1.

- Mahama, E. S. (2012). Ghanaian English and its implications for academic writing: A case study of English on the Navrongo campus of university for development studies, Ghana. *Research on Humanities and Social Sciences*, 2(11), 56-63.
- Malhotra, N. K., & Birks, D. F. (2003). *Marketing research: An applied approach*. Pearson Education Limited.
- Maniam, M. (2010). The influence of first language grammar (L1) on the English language (L2) writing of Tamil School students: A case study from Malaysia. *Language in India*, 10, 1-209.
- Martinez, L. (2020). Ambiguity and errors in student writing: A case study. *Journal of Language and Literacy Education*, 35(1), 56-72.
- Masda, S. S. (2007). How Ambiguous is the Structural. *Journal of Lingua Cultura*, 1(2): 99-104.
- McNamara, J. F. (1994). Surveys and experiments in education research. Technomic Publishing Company, Inc.
- Merriam, S. B. (2009). *Qualitative research: A guide to design and implementation*. San Francisco: John Wiley and Sons.
- Miles, M. B. & Huberman, A. M. (1994). *Qualitative data analysis (2nd ed.)*. Sage Publication.
- Ministry Of Education. (2013). Education sector performance report. MOE.
- Ministry of Education. (2022). Education sector performance report. MOE.
- Mireku-Gyimah, P. B. (2008). Do students of mining and allied engineering programmes have any problems in English? *Ghana Mining Journal*, 10, 48-62.
- Mitchell, D. C., Cuetos, F., Corley, M., & Brysbaert, M. (1995). Exposure-based models of ambiguity resolution. *Journal of Memory and Language*, 34(4), 521-542.
- Miyao, Y. (2006). From linguistic theory to syntactic analysis: Corpusoriented grammar development and feature forest model. (Unpublished PhD Thesis), University of Tokyo.

- Mohammed, I., and Amponsah, O. (2018). Predominant factors contributing to low reading abilities of pupils at Elsie Lund Basic School in the Tamale Metropolis, Ghana. *African Educational Research Journal*, 6(4): 273-278.
- Morris, J. M. (1984). Phonics 44 for initial literacy in English. *United Kingdom Reading* Association, 18(1), 13-24.
- Morrison, F. J., & Cooney, R. R. (2001). Parenting and academic achievement: Multiple paths to early literacy. In J. G. Borkowski, S. L. Ramey, & M. Bristol-Power (Eds.), Parenting and the child's world: Influences on academic,intellectual, and socio-emotional development. (pp. 141–160). Mahwah, NJ: Lawrence Erlbaum.
- Morrow, L. M. (1995). *A survey of family literacy*. Newark, DE: International Reading Association.
- Muhsin, M. A. (2016). Analysing the students errors in using simple present (A case study at Junior High School in Makassar). Pacific Science Review B: Humanities and Social Sciences, 2(3), 81-87.
- Mule, K. (2012). Types and Cause of Reading Difficulties Affecting the Reading OF English Language: A Case of Grade 4 Learners in Selected Schools in Ogong Circuit of Nambia. Unpublished MBA Thesis, University of Namibia.
- Murayama, K. (2007). A reward-learning framework of knowledge acquisition: an integrated account of curiosity, interest, and intrinsic-extrinsic rewards. *Psychological Review*, 12(2), 21-35.
- N'Namdi, K. (2005). *Guide to Teaching Reading at primary school Level*. France: UNESCO.
- Nagy, W. E. & Herman, P. A. (1987). Learning Word Meanings from Context during Normal Reading during Normal Reading. *American Educational Research Journal*, 24(2), 237-270.
- Nagy, W., & Anderson, R. C. (1984). How many words are there in printed school English? *Reading Research Quarterly*, 19(3), 304-330.

- Nalisa, T., Chataa, B. M., & Maemeko, E. L. (2019). Causes of Reading Difficulties in English 2nd Language in Grade 4 at a School in Katima Mulilo Circuit . *European Journal of English Language and Literature Studies*, 7(5), pp.27-35.
- National Endowment for the Arts. (2004). NEA report: Reading at risk. *Arts Education Policy Review*, 106 (1), 38-39.
- National Reading Panel. (2000). Teaching Children to Read: An Evidence-Based Assessment of the Scientific Research Literature on Reading and its Implications for Reading Instruction. Available at http: //www.nichd.nih.gov/publications/nrp/report.cfm.
- Nehafo, R. H. (2011). Investigating Reading Difficulties in English Second Language of Grade 3 Learners in One Primary School in the Khomas Education Region of Nambia. Unpublished MBA Thesis, University of Nambia.
- Nergis, A. (2013). Exploring the factors that affect reading comprehension of EAP learners. *Journal of English for Academic Purposes*, 12(1), 1-9.
- Nezami, S. A. (2012). A critical study of comprehension strategies and general problems in reading skill faced by Arab EFL learners with special reference to Najran University in Saudi Arabia. *International Journal Social Science and Education*, 2(3), 306-316.
- Ngure, W., Mwoma, T., & Buna, Y. (2019). Investigating the Levels of Reading Skills Among Grade Three Pupils in Nairobi County, Kenya. *European Journal of Education Studies*, 6(1): 197-209.
- Nguyen, L. V. (2016). Ambiguity in Student Writing: An Analysis of Lexical Ambiguity in Argumentative Essays. *Journal of Writing Research*, 8(3), 431-449.
- Nicol, J. L., & Swinney, D. A. (1989). The role of structure in coreference assignment during sentence comprehension. *Journal of Psycholinguistic Research*, 18(1), 5-20.
- Nkwi P. N., Nyamongo I. K., & Ryan G. W. (2001). Field research into socio-cultural issues: Methodological guidelines. International Center for Applied Social Sciences, Research, and Training.

- Nor, N. M., & Rashid, R. A. (2018). A review of theoretical perspectives on language learning and acquisition. *Kasetsart Journal of Social Sciences*, 39(1), 161-167.
- Norlander, I. (2012). Grammatical errors in written production by Swedish learners of English - A quantitative study of concord errors and selected verb errors in 35 national tests for the B-Course of English (Student essay). SPL bachelor's thesis in English, 48.
- Norris, J., & Ortega, L. (2000). Effectiveness of L2 Instruction: A Research Synthesis and Quantitative Meta-Analysis. *Language Learning*, 50, 417-528. http://dx.doi.org/10.1111/0023-8333.00136.
- Norton, B., & Toohey, K. (2011). Identity, language learning, and social change. Language Teaching, 44(04), 412-446. doi:10.1017/S0261444811000309.
- Nssien, F. U. (2007). "Reading Habits and Skills" In F. E. Etim and F. U. Nssien (eds.) Information Literacy for Library. Search, Uyo: Abaam Publishing, pp 90–105.
- Nunan, D. (1991). Language Teaching Methodology. Hertfordshire: Prentice Hall International.
- Nurjanah, R. L. (2018). The Analysis on Students' Difficulties in Doing Reading Comprehension Final Test. *Metathesis: Journal of English Language Literature* and Teaching, 2(2), pp. 253 - 264.
- Nuttall, C. (2000). *Teaching reading skills in a foreign language*. Richard Clay Ltd, Bungay, Suffolk.
- Nyarko, K., Kugbey, N., Kofi, C. C., Cole, Y. A., & Adentwi, K. I. (2018). English Reading Proficiency and Academic Performance Among Lower Primary School Children in Ghana. *SAGE Open*, 1–10.
- Obada, A. M. (2011). *Reading culture in Nigeria: A librarian's perspective*. Covena House Publishers, 53.
- Ogunrombi, S. A. & Adio, G. (1995). Factors affecting reading habits of secondary school students. *Library Review*, 44(4): 50 57.

Ogunsiji, A., & Sunday, A. (2005). Grammatical units. GSP.

- Ojanen, E. (2007). Sewero la-ma-u: A phonetic approach to literacy teaching in Zambia. Unpublished master's thesis, University of Jyvaskyla, Finland.
- Okebukola, F. O. (2006). *Reading: key to lifelong development*. The National Library of Nigeria.
- Okwilagwe, O. A. (1998). A New Approach to Reading Comprehension and Summary for schools and colleges. Horden Publishes (Nig) Ltd.
- Okyere, B., Ampadu, E., & Osei-Owusu, B. (2017). Reading comprehension assessment instruments used at lower primary level in Ghana: Do they match curriculum expectations? *Journal of Education and Practice*, 8(19), 1-10.
- O'Leary, Z. (2004). The Essential Guide to Doing Research. SAGE Publications Ltd.
- Oludare, O. J. (2016). Concord errors in postgraduates' dissertations and thesis. International Journal of English Language and Literature Studies, 5(1), 1-7.
- Oluwole, D. A. (2008). The impact of mother tongue on students' achievement in English language in junior secondary certificate examination in western Nigeria. *Journal of Social Science*, 17(1), 41-49.
- Omaggio, A. H. (1993). Research in language learning: principles, processes, and prospects (ACTFL Foreign Language Education Series). (pp. 45-46). National Textbook Company.
- Omotese, E. F. (2012). A Comparison of errors of concord students often conflict. *Elixir International Journal*, 49, 9862 – 9864.
- Oni, F. (2012). Wording the Gloom of an African Democracy: Lexical Choices and Cognition in Nigeria's President Olusegun Obasanjo's Inaugural Speeches. *Journal of English Studies*, 8, 31-48.
- Onovughe, O. G. (2012). Internet use and reading habits of higher institution students. Journal ofEmerging Trends in Educational Research and Policy Studies, 3 (1): 11-15.
- Onyejiaku, F. O. (2010). Techniques of effective study: A manual for students in colleges and universities. Wusen Press.
- Oriaku, R. (2006). The basic principles of the sentence. Postgraduate School.

- Oriogu, C. D., Esoname, S. R., Oriogu-Ogbuiyi, D. C., & Udoaku, O. S. (2017). Effect of Reading Habits on the Academic Performance of Students: A Case Study of the Students of Afe Babalola University, Ado-Ekiti, Ekiti State. *Teacher Education and Curriculum Studies*, 2(5): 74-80.
- Owu-Ewie, C & Williams, R. (2017). Grammatical and Lexical Errors in Students' English Composition Writing: The Case of Three Senior High Schools (SHS) in the Central Region of Ghana. *Sino-US English Teaching*, Vol. 14, No. 8, 463-482.
- Owu-Ewie, C. (2014). Readability of Comprehension Passages in Junior High School (JHS) English Textbooks in Ghana . *Ghana Journal of Linguistics*, 3(2): 35-68.
- Owusu-Acheaw, M. (2014). "Reading Habits Among Students and its Efect on Academic Performance: A Study of Students of Koforidua Polytechnic". *Library Philosophy and Practice (e-journal)*, Paper 1130.
- Owusu-Ansah, L. K. (1997). Nativisation and the Maintenance of Standards in Non-Native Varieties of English. Black Mask Publishers.
- Owusu-Boateng, F. (2008). Identifying Problems of Paragraph in the written Essays of Senior High School Students. Legon, Ghana: Unpublished M.A.Thesis, Department of Linguistics, University of Ghana.
- Paaku, V. E. (2008). Factors accounting for poor performance in basic education certificate examination in some selected junior secondary schools in Ajumako Enyan Essiam District. (Masters Dissertation), University of Education, Winneba.
- Parson, G. (1985). Hand in Hand: the Writing Process and the Microcomputer, Two Revolutions in the Teaching of Writing. A Manual for Secondary Teachers. Juneau, AK: Alaska State Department of Education (ED 264 - 598).
- Pateda, M. (2001). Semantik Leksikal. Jakarta: PT Rineka Cipta.
- Paulson, E. J. (2006). Self-selected reading for enjoyment as a college developmental reading approach. *Journal of College Reading and Learning*, 36 (2): 51-58.
- Perfetti, C. A., & Hart, L. (2002). The lexical quality hypothesis. Precis, 6, 30-40.

- Pham, L. H. (2004). University classrooms in Vietnam: Contesting the stereotypes. *ELT Journal*, 58(1), 50-57.
- Pramitasih, A. (2012). A Study On The Ambiguity Found In English Exercises of Vocational School Student's Exercise Books. Surakrta: Doctoral dissertation, Universitas Muhammadiyah Surakrta.
- Pratiwi, A. P. (2015). Interlingual and intralingual errors of writing narrative text made by Junior High School and Senior High School students. Muhammadiyah University of Surakarta.
- Pressley, M., & Afflerbach, P. (1995). Verbal protocols of reading: The nature of constructively responsive reading. Lawrence Erlbaum Associates.
- Pritchard, R. M., & Honeycutt, R. L. (2005). The effects of reading and writing upon thinking critically. *Journal of College Reading and Learning*, 35(2), 25-39.
- Pylkkänen, L., & Marantz, A. (2003). Tracking the time course of word recognition with MEG. *Trends in Cognitive Sciences*, 7(5), 187-189.
- Qrqez, M. & Ab-Rashid, R. (2017). Reading Comprehension Difficulties among EFL Learners: The Case of First and Second -Year Students at Yarmouk University in Jordan. *Arab World English Journal*, 8(3), Pp. 421-431.
- Qrqez, M., & Radzuwan, A. R. (2019). The Sources of Reading Comprehension Difficulties Among Saudi EFL Learners. *Trends in Social Sciences*, 1(1) 7-16.
- Quirk, R. (2010). A comprehensive grammar of the English language. Pearson Education.
- Quirk, R., Greenbaum, S., Leech, G. & Svartvik, J. (2005). *A comprehensive grammar* of the English language. Longman.
- Rahimi, A. R & Talepasan, I. E. (2012). Exploring EFL Learners Reading Comprehension Problems in Reading ESP Texts. *Sino-US English Teaching*, 9(3): 982-987.
- Rajab, H., & Al-Sadi, A. (2015). An Empirical Study of Reading Habits and Interests of Saudi University EFL Learners. *International Journal of Linguistics*, 7(2): 1-17.

- Rao, Z. (2007). Training in brainstorming and developing writing skills. *ELT Journal*, 61 (2), 100-106.
- Rataj, K. (2020). 10 electrophysiology of semantic violations and lexical ambiguity resolution in bilingual sentence processing. *Bilingual Lexical Ambiguity Resolution*, 250.
- Rector, M. A., Nehm, R. H., and Pearl, D. (2013). Learning the language of evolution: lexical ambiguity and word meaning in student explanations. *Research in Science Education*, 43(3): 1107-1133.
- Reeve, D. B. (2009). The learning leader/the value of culture. *Educational Leadership*, 66(7), 87-89.
- Reid, J. (1998). 'Eye'' learners and 'ear'' learners: Identifying the language needs of international students and U.S. resident writers. In P. Byrd and J. M. Reid (Eds.), Grammar in the composition classroom: Essays on teaching ESL for college-bound students (pp. 3-17). Boston: Heinle & Heinle.
- Reid, J. and Donaldson, H. (1977). *Reading Problems and Practices, (2nd ed.)*. London: Richard Clay Limited.
- Richards, J. C., and Schmidt, R. (2002). *Dictionary of Language Teaching & Applied Linguistics*. London : Pearson Education Limited.
- Rodd, J. M., Gaskell, M. G., and Marslen-Wilson,W. D. (2002). Making sense of semantic ambiguity: Semantic competition in lexical access. *Journal of Memory* and Language, 46: 245-266.
- Roth, R. M. (2002). Under achieving students and guidance. Houghton: Miffen Press.
- Roth, W. (2007). Vygotsky's neglegted legacy: Cultural- historical activity theory. *Review of Educational Research*, 77-186.
- Rumelhart, D. E. (1980). Towards an Interactive Model of Reading. (Dorinic S. ED) Attention and Performance (pp. Vol. 6, 573-603). New YorK: Academic Press.

- Rutzel, D. R. & Cooter, R. B. (2010). Developing the Literacy Instruction Knowledge Scale (LIKS): A Comprehensive Assessment of Primary Grade Teachers Knowledge of Reading and Instructions. The American Educational Research Association.
- Samuels, S. J., and Kamil, M. L. (1988). "Models of the Reading Process." In Carrell, Devine, and Eskey, eds. 22–36.
- Scott, D. (2009). Reading Educational Research and Policy. Routledge.
- Sennet, A. (2016). "Ambiguity" Stanford Encyclopedia of Philosophy (Spring Edition 2016 Edition).
- Setiani, R. (2019). Students' Ambiguity in Writing An Essay. Jurnal Elsa, 17 (1).
- Setiawan, R. G. (2014). Investigating ambiguity in efl student writing. *Repository* Universitas Kristen Satya Wacana, 1–26.
- Shanahan, T., & Shanahan, C. (2008). Teaching disciplinary literacy to adolescents: Rethinking content-area literacy. *Harvard Educational Review*, 78(1), 40-59.
- Shankweiler, D., & Fowler, A. E. (2004). Literacy, Reading Development, and Dyslexia. In Handbook of Developmental Cognitive Neuroscience (2nd ed) (pp. 607-628). MIT Press.
- Sheen, R. (2002). A Response to Lightbown's (2000) 'Anniversary article: Classroom SLA research and second language teaching'. *Applied linguistics*, 23(4), 519-528.
- Shen, L. (2006). Computer technology and college students' reading habits. *Chia-Nan Annual Bulletin*, 32, 559–572.
- Shiotsu, T., & Weir, C. J. (2007). The relative significance of syntactic knowledge and vocabulary breadth in the prediction of reading comprehension test performance. *Language Testing*, 24(1), 99-128.
- Shonkoff, J. P. and Phillips, D. A. (2000). From Neurons to Neighborhoods: The Science of Early Childhood Development. National Academies' Press.

Shuttleworth, M. (2008). Qualitative research design. Sage Publication.

- Silverman, D. (2010). *Doing qualitative research: a practical handbook, 2nd ed.* Sage Publication.
- Smith, F. (1994). Understanding Reading, 5th ed. Lawrence Erlbaum.
- Smith, J., & Johnson, L. (2015). Decoding Difficulties and Reading Comprehension: A Study on Slow Readers. *Journal of Reading Research*, 25(3), 112-126.
- Snow, C. E., & Uccelli, P. (2009). The challenge of academic language. In D. R. Olson,
 & N. Torrance (Eds.), The Cambridge handbook of literacy (pp. 112-133).
 Cambridge University Press.
- Sokeng, S. C. (2014). Grammatical errors of bilingual 1 francophone learners of English in the University of Yaounde I. *Theory and Practice in Language Studies*, 4(9), 1778-1785.
- Stageberg, N. C. (1966). Structural ambiguity: some sources. *The English Journal*, 55(5): 558-563.
- Stahl, S. A., & Fairbanks, M. M. (1986). The effects of vocabulary instruction: A model-based meta-analysis. *Review of Educational Research*, 56(1), 72-110.
- Stahl, S. A., & Nagy, W. E. (2006). *Teaching Word Meanings*. Lawrence Erlbaum Associates.
- Stake, R. E. (2004). Standards based and responsive evaluation. Sage Publications.
- Stanovich, K. E. (1980). "Toward an Interactive-Compensatory Model of IndividualDifferences in The Development of Reading Fluency". *Research Reading Quarterly*, 16(1): 32-71.
- Stotsky, S. (1983). Research on reading and writing relationship: A synthesis and suggested direction. *Language Arts*, 66(50), 627-642.
- Stubbs, M. (1983). Texts, corpora, and problems of interpretation: A response to Widdowson. Applied linguistics, 22(2), 149-172.
- Swan, M. (2005). Practical English usage. Oxford University Press.
- Swanson, H. L., & Jerman, O. (2007). The Influence of Working Memory on Reading Growth in Subgroups of Children with Reading Disabilities. *Journal of Experimental Child Psychology*, 96(4), 249-283.

- Swinney, D. A., & Cutler, A. (2000). The access and processing of idiomatic expressions. *Journal of Verbal Learning and Verbal Behavior*, 18(5), 523-534.
- Tafazoli, D. & Kargozar, H. R. (2015). An investigation of factors affecting Iranian EFL learners' writing accuracy: syntactic, lexical, or other factors. *International Journal of Research Studies in Language Learning*, 5(3), 51-60.
- Taha, A. K. (1983). "Types of Syntactic Ambiguity in English". *IRAL*, Vol. XXI/4. 251–266.
- Takala, A. (2016). Grammar teaching methods in EFL lessons: Factors to consider when making instructional decisions. (Unpublished Masters thesis), University of Jyväskylä.
- Tan, H. (2001). A study of EFL learners' writing errors and instructional strategies.
- Tannor, L. L. (2014). A guide to writing dissertations. Blessed Publication.
- Teevno, R. A., and Raisani, R. B. (2017). English Reading Strategies and their Impact on Students' Performance in Reading Comprehension. *Journal of Education & Social Sciences*, 5(2): 152-166.
- Thakur, D. (2002). A Handbook of English Grammar and Usage. Bharati Bhawan.
- The National Endowment for the Arts. (2004). *To Read or Not To Read A Question of National Consequence*. Don Ball Publication.
- Thomson, A. J., & Martinet, A. V. (1997). *A practical English grammar*. Oxford University Press.
- Tierney, R. J., and P. D. Pearson. (1994). "Learning to learn from a text: A Framework forImproving Classroom Practice". *In Rudell, Ruddell, and Singer, eds*, (pp. 496–513).
- Totto, P. M., & Ramos, A. B. (2021). Reading and Writing Performance of Senior High School Students. *International Journal of English Language Studies*, 3(9):09-22. DOI:10.32996/ijels.2021.3.9.2.
- Trang, N. H. (2021). Reading Comprehension Difficulties Encountered by First Year English-Majored Students. *International Journal of Education, Learning and Development*, 9(9), pp.51-67.

- Truscott, J. (1996). The case against grammar correction in L2 writing classes. Language Learning Journal, 46, 327-369.
- Tsuchida, T. (2011). A descriptive enquiry into subject-verb concord in English existential constructions. Digital Hollywood University.
- Usher, E. (2009). Sources of Middle School Students' Self-Efficacy in Mathematics. A Qualitative Investigation. *American Educational Research Journal*, 46, 275-314.
- Vilenius-Tuohimaa, P. M., Aunola, K. & Nurmi, J. (2008). The association between mathematical word problems and reading comprehension. *Educational Psychology*, 28(4), 409-426.
- Walczak, A. & Geranpayeh, H. D. (2015). Is there a gender divide in English language performance? *Springer*, 139-190.
- Weiner, B. (2000). Intrapersonal and interpersonal theories of motivation from an attributional perspective. *Educational Psychology Review*, 12(1), 1-14.
- Weisman, H. M. (2007). Basic Technical Writing. Ohio: Charless E. Meririll.
- West African Examination Council. (2022). English Language Examination Syllabus.
- Wigfield, A., Guthrie, J. T., Tonks, S., & Perencevich, K. C. (2004). Children's motivation forreading: Domain specificity and instructional influences. *Journal* of Educational Research, 97 (6): 299-309.
- Wilson, K. & Bhamjee, Y. (2007). Promoting Literacy in Secondary Content Area Classrooms with Biography Projects. *Journal of Adolescent & Adult Literacy*, 40(4): 270-278.
- Wilson, L. (2016). Anderson and Krathwohl Bloom's Taxonomy Revised Understanding the New Version of Bloom's Taxonomy.
- Wiredu, K. (1996). Cultural universals and particulars: An African perspective. Indiana University Press.
- Zainal, Z., & Husin, S. H. B. M. (2011). A Study On The Effects Of Reading On Writing Performance Among Faculty Of Civil Engineering Students. *TESOL Quarterly*, 1-10.

University of Education,Winneba http://ir.uew.edu.gh

- Zamel, V. (1992). Writing one's way into reading. TESOL Quarterly, 26(3), 463-485.
- Zhang, D. (2012). Vocabulary and grammar knowledge in second language reading comprehension: A structural equation modeling study. *The Modern Language Journal*, 96(4), 558-575.



University of Education, Winneba http://ir.uew.edu.gh

APPENDIX

UNIVERSITY OF EDUCATION, WINNEBA FACULTY OF FOREIGN LANGUAGES EDUCATION DEPARTMENT OF APPLIED LINGUISTICS INTERVIEW GUIDE FOR ENGLISH TEACHERS

Dear Sir/Madam

Introduction

This study seeks to investigate "the impact of reading skills on students' writing using students of Amuman M/A Basic School as a case". You are kindly requested to volunteer information by responding to this interview. You are however assured that information provided to complete this interview would be treated with the strictest confidentiality.

QUESTIONS

- 1. Can you tell me what problems you face when trying to understand what you read?
- 2. How well do you understand and use new words you come across while reading?
- 3. Do you find it difficult to read smoothly and at the right speed? Why?
- 4. Do you have trouble connecting what you read to what you already know or have experienced?
- 5. Are there any types of texts or subjects that you find hard to understand? Why?

- 6. What do you do when you come across unfamiliar or difficult words while reading?
- 7. What kind of help or resources do you think would improve your reading comprehension skills? Is there something specific you would like assistance with?

